# Modelling and Simulation of Traffic signal control: A case study of Islamabad city



By

## Fatima Ghaffar - 00000273507

### Supervisor

# Dr. Muhammad Tariq Saeed

A dissertation submitted to National University of Sciences & Technology in partial fulfillment of the requirement for the degree of MASTER OF SCIENCE

in

Computational Science and Engineering

Research Center for Modeling and Simulation (RCMS) National University of Sciences and Technology (NUST) Islamabad, Pakistan

September,2021

# **Declaration of Authorship**

I Fatima Ghaffar, declare that this thesis titled "Modelling and Simulation of Traffic signal control: A case study of Islamabad city" and the work presented in it are my own. I confirm that:

- This work was done wholly or mainly while in candidature for a research degree at NUST.
- Where any part of this thesis has previously been submitted for a degree or any other qualification at NUST or any other institution, this has been clearly stated.
- Where I have consulted the published work of others, this is always clearly attributed.
- Where I have quoted from the work of others, the source is always given. With the exception of such quotations, this thesis is entirely my own work.
- I have acknowledged all main sources of help.
- Where the thesis is based on work done by myself jointly with others, I have made clear exactly what was done by others and what I have contributed myself.

Signed:

Date:

# **Copyright Statement**

- Copyright in text of this thesis rests with the student author. Copies (by any process) either in full, or of extracts, may be made only in accordance with instructions given by the author and lodged in the Library of NUST Research centre of modelling and simulation (RCMS). Details may be obtained by the Librarian. This page must form part of any such copies made. Further copies (by any process) may not be made without the permission (in writing) of the author.
- The ownership of any intellectual property rights which may be described in this thesis is vested in NUST Research centre of modelling and simulation, subject to any prior agreement to the contrary, and may not be made available for use by third parties without the written permission of the RCMS, which will prescribe the terms and conditions of any such agreement.
- Further information on the conditions under which disclosures and exploitation may take place is available from the Library of NUST Research centre for modelling and simulation, Islamabad.

# Dedication

To my Parents and Siblings

# Acknowledgements

All praise to ALLAH the Almighty, the creator of this universe, who has blessed mewith the sense of thankfulness, humbleness and love. The owner of crown, the repellerof Affliction and disease, the master of Madina, Prophet Muhammad (SAW). I praythat He always keeps me under your supervision.

I an thankful to the people who have contributed in my research, inspired me with theirideas and devotion for country and guided me during my stay at NUST. Especially, I owemy gratitude to my supervisor **Dr. Muhammad Tariq Saeed** for continuously showingconfidence and trust in me. I am also thankful to **Dr. Rizwan Riaz**, **Dr. Adnan Maqsood** and **Dr. Mian Ilyas Ahmad** for their endless support. I would extend my gratitude toall my friends, seniors and juniors especially my classmates.Special thanks to my parents who made me the person who I am today. I can't thankyou enough for the sacrifices you have made to support me.

### Abstract

Traffic congestion is a major issue in many modern cities around the world. Over many years, different solutions have been proposed and tested to solve the problem. One such solution is to build a new traffic infrastructure however this proves to be expensive and may not be possible in some places. Another solution is to optimize the existing traffic network by smartly controlling traffic lights at intersections, to ensure that vehicles move smoothly along transportation routes. Analysing the situation in Pakistan, the current traffic lights in Islamabad city, which are installed across the city, are insufficient to address the congestion issues since they have particular pre-determined times for green and red lights. This results in large queues leading to disruptions in the daily lives of passengers. In this research, we used fixed and actuated traffic light control methods to optimize traffic signal programs generated by SUMO, which is an open-source microscopic simulator. By using delay time and queue length as the parameters to reduce the congestion, one single intersection has been modelled in SUMO.

Actuated traffic light control has shown a strong potential to cater to traffic fluctuations to achieve desired objectives and effectively reduced traffic congestion by adjusting traffic signal plans. To test the efficacy of the used strategy, one single intersection has been used which is controlled by an Intelligent traffic light control plan (ITCS). Further, actuated traffic light control was found to reduce intersection delays by up to 27% relative to fixed traffic light control which can be further improved by using adaptive traffic light control strategies based on machine learning algorithms. Moreover, Fluctuations in traffic flow have been catered better by actuated traffic light control than fixed traffic light plan.

# Contents

De	eclara	tion of	Authorship				i
Co	opyri	ght Stat	tement				ii
De	edicat	tion					iii
Ac	eknov	vledger	nents				iv
Ał	ostrac	t					v
Li	st of ]	Figures					ix
Li	st of '	Tables					xi
Ał	obrev	iations					xii
1	Intr	oductio	on				1
	1.1	Backg	round				1
	1.2	Intellig	gent Traffic Control System (ITCS)		•		2
	1.3	Proble	m Statement		•		3
	1.4	Identif	ied challenges		•		6
		1.4.1	Conduct of commuter		•		6
		1.4.2	Educational Institute		•		6
		1.4.3	Construction/Renovation of Roads and Pavements		•		6
		1.4.4	Traffic Lights (Signals)	•••	•	•	7
	1.5	Motiva	ation	• •	•	•	7
	1.6	Resear	ch Objective	• •	•	•	8
	1.7	Organ	ization of the Thesis	•••	•	•	9
			Chapter 1: Introduction	• •	•	•	9
			Chapter 2: Literature Review	• •	•	•	10
			Chapter 3: Problem Formulation	• •	•	•	10
			Chapter 4: Results and Discussion	• •	•	•	10
			Chapter 5: Conclusion and Future Recommendations	• •	•	•	10

2 Literature Review

	2.1	Simula	ation
		2.1.1	WHY SIMULATE? 12
		2.1.2	TYPES OF TRAFFIC SIMULATION 13
		2.1.3	Macroscopic (macro) models
		2.1.4	Microscopic (micro) models
		2.1.5	Mesoscopic (meso) models
	2.2	Traffic	e Signal
		2.2.1	Communications
		2.2.2	Maritime Navigation
		2.2.3	Goal of Traffic Signal Control 16
		2.2.4	Highway signals
	2.3	Traffic	c Control Settings Overview 17
		2.3.1	Design of Signal Plan
		2.3.2	Timing of Signal Plan 18
	2.4	Differe	ent techniques used for traffic signal control
		2.4.1	Manual traffic control management
		2.4.2	Automatic Traffic Management Technique
		2.4.3	Image Processing based Intelligent Traffic Management Technique 22
		2.4.4	Traffic Management system using Wireless Technologies 22
		2.4.5	IRIS (Intelligent Roadway Information System)
	2.5	Intelli	gent Transportation Systems
			Pre-timed:
			Actuated:
			Adaptive:
	2.6	Fixed	time Control (Pre-Timed Control)
		2.6.1	Webster Method25
	2.7	Actuat	ted Time Control
		2.7.1	Traffic Actuated controller 27
		2.7.2	Traffic Simulator 28
2	Mat	hadala	
3		Madal	gy 29
	5.1	2 1 1	Formulation
		3.1.1	Pour Phase Signals
		3.1.2	Noteconvert
		5.1.5 2.1.4	Debreement 22
		3.1.4	Polyconvert
		3.1.5	NetLait
		3.1.0	IfaCI
		3.1.7	Simulation Definition
		J.1.0 2 1 0	Cor Following Model
	2 0	3.1.9 Тисфа	Car ronowing model
	3.2		The traffic light control Algorithm
		5.2.1	Fixed traffic light control
		5.Z.Z	Fixed traffic light control    43      Astrophysical traffic light control    44
		3.2.3	Actuated traffic light control

	4.1	Fixed-Time control (Pre-Timed control)	50
	4.2	Actuated time traffic control	53
	4.3	Comparison of all Configurations	56
5	Con	clusions and Recommendations	57
	5.1	Conclusions	57

#### Bibliography

# **List of Figures**

1.1	Open Street Map of Islamabad Srinagar Highway	5			
1.2	General Methodology Flow Chart 9				
2.1	Application Area and Benefits of Microscopic Traffic Simulation Models [55] .	12			
2.2	Green interval	19			
2.3	Yellow interval	20			
2.4	All-Red Clearance Interval	21			
2.5	Green time extension	27			
3.1	Description of Phases on an intersection	31			
3.2	Detailed Description of Phases on an intersection	31			
3.3	Improved network	33			
3.4	Original OpenStreetMap network of Srinagar Highway	38			
3.5	Road Network using Open street map with link of .net files	38			
3.6	Traffic light control flow chart	42			
3.7	Green lines showing fixed Signal plan of Srinagar highway facing Islamabad to				
	Rawalpindi having straight and right turn directions	44			
3.8	Green lines showing fixed Signal plan of Service road facing NUST to G11 hav-				
	ing straight and right turn directions	44			
3.9	Green lines showing Fixed Signal plan of Srinagar highway facing Rawalpindi				
	to Islamabad having straight and right turn directions	45			
3.10	Green lines showing Fixed Signal plan of Service road facing G11 to NUST				
	having straight and right turn directions	45			
3.11	Green lines showing Actuated Signal plan of Srinagar highway facing Islam-				
	abad to Rawalpindi having straight and right turn directions	46			
3.12	Green lines showing Actuated Signal plan of Service road facing NUST to G11				
	having straight and right turn directions	47			
3.13	Green lines showing Actuated Signal plan of Srinagar highway facing Rawalpindi				
	to Islamabad having straight and right turn directions	47			
3.14	Green lines showing Actuated Signal plan of Service road facing G11 to NUST				
	having straight and right turn directions	48			
3.15	Flow chart of Methodology	48			
4.1	Configuration of cars in fixed traffic light control	51			
4.2	Configuration of equally distributed traffic in fixed traffic light control	51			
4.3	Configuration of 2 lanes fixed for heavy traffic in fixed traffic light control	52			
4.4	Comparison of all four sides of an intersection with taking different vehicular				
	input	53			
4.5	Configuration of Cars as vehicular input in Actuated traffic light control	54			

4.6 Configuration of Equally distributed traffic as vehicular input in Actuated tra		
fic light control	55	
Configuration of 2 lanes fixed for heavy traffic as vehicular input in Actuated		
traffic light control	55	
Comparison of all configuration for actuated traffic light control	56	
comparison of four phases at an intersection of G11	56	
	Configuration of Equally distributed traffic as vehicular input in Actuated traffic light control	

# List of Tables

3.1	Simulation files description	32
3.2	Traffic generation in different traffic scenarios	35
3.3	Traffic generation in different traffic scenarios	35
3.4	Low and High traffic scenario probability distribution	36
3.5	EW traffic scenario probability distribution	36
3.6	NS traffic scenario probability distribution	36
3.7	Population parameters used to generate vehicular input	37
3.8	Duration plan for Fixed Traffic light control	43
3.9	Duration plan for Actuated Traffic light control	46

# Abbreviations

ITCS	Intelligent Traffic Control System
DSRC	Dedicated Short Range Communication
IoT	Internet of Things
RFID	Radio Frequency Identification System
UWB	Ultra Wide Band
SCATS	Sydney Coordinated Adaptive Traffic System
SCOOT	Split Cycle Offset Optimisation Technique
OPAC	Optimization Policies for Adaptive Control
RHODES	Real-Time Hierarchical Optimized Distributed Effective System
SUMO	Simulation of Urban MObility
ATMS	Advanced Traffic Management System
IRIS	Intelligent Roadway Information System
GPL	General Public License
SPOT	System for Priority and Optimisation of Traffic
OSM	Open Street Map
minDur	minimum Duration
maxDur	maximum Duration

TraCI Traffic Control Interface

### Chapter 1

# Introduction

#### 1.1 Background

In the 21st century, with rapidly increasing world's population, the urban population is projected to rise significantly. It is therefore essential to have managed urban infrastructure to cope with this rise effectively. Modern society depends on the systems of transport for the transportation of people, goods and services to ensure that vehicles can travel efficiently from point of origin to point of destination [18].

More vehicles on road cause traffic congestion that emerged as a very serious issue in populated cities that needs to be dealt appropriately. Traffic congestion leads to problems like fuel consumption,  $CO_2$  emission and wastage of time. The large number of vehicles, improper traffic infrastructure and inefficient traffic control system are the main reasons for traffic congestion. It is impossible to stop people from buying cars and to start building new traffic infrastructure that might cause a huge burden on country's economy. So, the easy way to solve this problem is to optimize the traffic signal timing at intersection.

The demand for road infrastructure has also risen beyond its capacity due to an increase in population and ownership of vehicles which results in congestion, travel delays and excessive emissions of vehicles. Up till now, road infrastructure improvements have been used to reduce traffic congestion [15] [52]. Even then, having strict constraints on financial resources and physical conditions, also environmental concerns, have amplified the need for alternative ways to minimize traffic congestion. We can only proceed with enhancing the utilisation of existing

transportation facilities to discover the right signal intersection control algorithm reducing land resources.

Traffic congestion is a challenging phenomenon that hits the everyday life of many people around the world in many countries [14]. At the same time, traffic congestion would also lead to excessive loss of resources and environmental pollution and increase the risk of traffic accidents. Traffic junctions play a significant influence in determining the congestion condition of the road network [31]. Most traffic junctions around the world use pre-defined fixed signal timings which are not equipped to track actual traffic [56]. Recent studies suggest hand-crafted rules based on actual traffic data. However, those rules cannot be dynamically modified according to real traffic data due to pre-defined control in nature. While such a technique is simple to enforce, it does not take into account real traffic conditions and can lead to more congestion.

In this chapter, we will briefly discuss the concept of Intelligent traffic control systems and traffic modelling. After that, we will introduce the problem and discuss the motivation, and finally, we will review the whole thesis.

#### 1.2 Intelligent Traffic Control System (ITCS)

Over the past few years, several Intelligent Traffic Control (ITS) systems [34] have been progressed and illustrated to be one of the most efficient methods to minimize traffic volume. Video cameras or loop detectors are used to get real-time traffic information which can be used to configure the traffic light split-cycle accordingly. Unfortunately, such intelligent traffic signal control systems are costly and thus only operate on the few interchanges of the United States, Europe and Asia.

The most frequently used control systems are pre-timed and actuated traffic signal control systems. Pre-timed signal control is best suited for fixed positioning strategies but not designed to cater to abrupt changes inflow of traffic. Even though simple and do not require professional skills but time-consuming. On the other hand, Actuated signal control response to modification of green time (minimum green to maximum green) varies based on traffic volume. Although it has been shown that actuated signal control can perform better than pre-timed in certain situations but does not give any real-time optimization for fluctuation in traffic conditions. As a result, actuated signal control is not flexible and adaptable to traffic fluctuations and could result in long queues in more than one intersections [7].

Recently, many more economical techniques have been introduced to deploy the ITSC systems through the use of Dedicated Short Range Communication (DSRC) technology which is a significantly less expensive technology for the detection of the vehicle on junctions. However, only a limited portion of vehicles would be fitted with DSRC radios at the early stages of deployment. Even now the rise of the Internet of things (IoT) has developed emerging technologies for tracking vehicles for ITSC. Other than DSRC, the innovations which are more cost-effective than conventional loop detectors or cameras include RFID, Bluetooth, Ultra-Wide Band (UWB), ZigBee, and even cellphone applications such as Google Map. The performance of these devices is better than loop detectors which only detect the existence of vehicles. However, the systems that have not fitted with mentioned communication devices would unable to detect the vehicle [21].

Generally, centralized real-time traffic system like SCATS and SCOOT are broadly utilized traffic signal control systems and dynamic optimization of the system to obtain signal settings represented by OPAC and RHODES.

Implementation of smart traffic management systems is one of the most important factors in the design of modern cities. A traffic management system's main goal is to minimise traffic congestion, which is one of the major issues in mega cities. Effective urban traffic control leads to saving time and finances also the reduction of  $CO_2$  emissions into the atmosphere. Improving traffic conditions will increase city productivity, boost the economy and make people's everyday life easier [68] [10].

So computers can learn to make a reasonable decision by interacting with the environment. With the advancement of the internet and wireless communications, data processing has become more convenient and quicker than ever. The fair use of different traffic data is also crucial to lessening traffic congestion.

#### 1.3 Problem Statement

In developing countries, traffic infrastructure is in a progressive state and changes every day. Old road maps are being changed by establishing new roads. This causes a tremendous change in traffic flow on different routes and leads to traffic congestion problem. The most common approach used to solve congestion problem is to predefine the green phase duration according to historical traffic demands. It fails most of time because it does not consider the fluctuation in traffic demand. Other solution uses the sensors like inductive loop detector to control the traffic signal. Inductive loop detector has poor detection for smaller vehicle and can be damaged by road deterioration. A system is needed that not just controls the traffic flow at intersection but also adopts new control policy according to traffic behavior.

Islamabad is the capital city of Pakistan located in the Potohar Plateau in the north west of Pakistan with an elevation of 500 meters above sea level. Islamabad is located at latitude of 33.71° North and longitude of 73.1° East, and is bordered by Margalla hills on one side (North) and plains of an adjunct province (Punjab) on the other side. The city of Islamabad is spread over an area of 906 km2 with a reported population exceeding one million [59], [71]. Having the status of federal and administrative capital, Islamabad city has a national importance as well. This significance of Islamabad has always invited new dwellers into the city that has resulted in a constant expansion of urban population [6].

One of the consequences of such a population increase is the massive rise in the transportation flux observed during the last decade [66]. Easy availability of passenger vehicles on lease from commercial banks has also contributed towards the increasing vehicle population which in turn has increased the air pollution in the city [23]. Figures for the last five years have also shown an average annual addition of 62,000 vehicles in the city's overall fleet.

Islamabad has a adjoining city, Rawalpindi, which is only a few kilometres away. These cities are linked, and in recent years, approximately 100,000 people drive everyday from Rawalpindi to Islamabad. The daily movement of these 100,000 people to the capital caused traffic congestion, and the absence of a local transportation network made it even most complicated for travellers. Pakistan, still in its early stage of growth, faces a lack of public transportation, such as metros and buses, which are important in any metropolitan city. Besides, a lot of traffic congestion, Pakistan is coping with an electricity shortage, with 35 and 25 per cent of the existing shortfall occurring during peak and off-peak days, respectively. Furthermore, since 1964, the inhabitant of this city has increased from 100,000 to 2 million [67].

In Islamabad, Pakistan's Capital Territory, the Srinagar Highway is a big east-west highway. The highway links the Islamabad International Airport in the west to the E-75 Expressway in the east, providing easy access to Islamabad. The highway is 25 kilometres long in total. There are five interchanges and the width ranges from three to five lanes. This research mainly focuses on the G11 intersection on Srinagar highway, G-11 (North Exit), Sector H-12 (NUST) (South Exit). G11 intersection is selected based on heavy traffic on both sides of the highway. Due to the main campus of an educational institute, nearly 5000 people daily commute through this highway. On the opposite side, the G11 sector provides a passage to all other sector and has a grand supermarket which also attracts people results in traffic congestion [33] [32].

G-11 intersection is four-way intersection. The east and west side of intersection have 5 lanes while north and south side of intersection has 3 lanes. The road approaching the intersection form north side has curvature. This intersection has separate road for left turn and does not required a traffic light.



FIGURE 1.1: Open Street Map of Islamabad Srinagar Highway

For major cities around the country the problems are worsening, although at a slower rate, but capacity is decreasing to absorb newcomer transportation, causing massive congestion. So, Major reasons for this growth are as follows:

- Leasing/car financing facilities provided by almost all the banks on low interest rates and on easy terms and conditions have contributed the most towards this end.
- Second major factor is domestic car manufacturing industry has also improved a lot in near past. Numbers of leading car manufacturers especially from Japan have started manufacturing their brands in Pakistan.
- Last decade has also seen tremendous improvement in road infrastructure especially the motorways. This has also encouraged people to buy new cars.
- Ever increasing Population

• Inadequate public transport facilities have made car maintenance a necessity rather than a luxury.

For major cities around the country the problems are worsening, although at a slower rate, but capacity is decreasing to absorb newcomer transportation, causing massive congestion.

#### 1.4 Identified challenges

#### 1.4.1 Conduct of commuter

Since Pakistan has much smaller cars than any other country, it causes more traffic congestion. Two small cars usually remain in a single lane and then switch lanes depending on the room they encounter, big vehicle drivers behind them become impatient and find it difficult to pass the smaller cars. While traffic wardens on-site play an important role in guiding commuters to keep lanes clear and often succeed, when they leave, the same situation occurs[69].

#### 1.4.2 Educational Institute

since Islamabad lacks a public transportation system such as buses or subways. Few people use the institute's transportation, which only carries 15 per cent to 20 per cent of the total student body, and the rest travel on their own. So there are two peak times for this type of congestion, one from 8:30 a.m. to 10:30 a.m. and the other from 4:30 p.m. to 8:30 p. m.

#### 1.4.3 Construction/Renovation of Roads and Pavements

This is a universal barrier to smooth traffic flow, however industrialized world have introduced regulations for the building of roads and sidewalks, as well as the repair and filling of sinkholes on a routine basis. Conversely, it has been noticed that during road-building in Pakistan, no such preventive steps are taken, leaving the condition in the hands of travellers. Many accidents occur during road construction. Typically, when travellers see a heap of concrete or construction material on the road, they apply abrupt breaks, and the car behind them does not respond as quickly, causing many to collide. This occurs most often during peak hours, as everyone insists on arriving at their destination on time and in the fastest manner possible. On Pakistani highways, it is common to practise to begin construction work without informing commuters, causing significant inconvenience [48].

#### **1.4.4 Traffic Lights (Signals)**

Many advanced steps are taken in order to enhance the functioning of traffic signals around the world. In many places, technology involving sensors is being used to enhance traffic flow and increase commuter comfort. In the current scenario, cities such as Islamabad are fitted with traffic signals, but the signal's reliability is very poor. The primary causes are a lack of preventative maintenance, electricity shortage, and defective lighting. When traffic signals fail due to an unplanned power outage, traffic is controlled by a traffic police; if the traffic person is unavailable, traffic navigates on its own. This is in contrast to the West, where the general rule is that when a traffic light ceases to function, drivers should recognise it as a STOP sign, and when they enter the junction, they should completely stop their cars, give everyone access, and then continue to cross the intersection.

In Pakistan, drivers roam freely on roads independent of specified lanes. This creates chaos and clogs up the traffic flow. When managing traffic manually, wardens often cause traffic congestion as they do not control traffic with regard to time, but rather transfer traffic based on the number of vehicles arriving from a single side. They allow about 2-3 minutes from that side, after which the remaining sides become congested, and one is only able to move after about 8-10 minutes. On the other hand, when traffic signals are operational, traffic on each side moves based on a timer set for them [77]. As a result of this operation, more cars are available for the subsequent traffic signal, major roads become congested, and commuters' average travel time nearly quadruples. During power outages, this is a common occurrence, and traffic wardens will often purposefully switch off traffic signals and control traffic themselves. Each day, the traffic department tries new approaches, and numerous traffic lights become inoperable as crossings are closed by hurdles to block traffic on both sides, forcing drivers to make a U-turn [26].

#### 1.5 Motivation

Simulation modeling is an increasingly popular and effective tool for analyzing and evaluating different alternatives/solutions. Most of the simulation models were developed in Europe and US. Model parameters are based upon their local conditions. If we want to use them in Pakistan, then these model parameters should be perform according to local traffic conditions. Although, most of the time in developed and advanced countries, driver behavior have similar algorithm, but when we are comparing these with third world countries it may differ significantly. Hence to use these micro simulation models in Pakistan, we must calibrate and validate these models.

#### 1.6 Research Objective

The main goal of this research is to investigate and minimize the congestion at the selected junction of Islamabad city by selecting different configurations based on the input of vehicles. The main objectives of this research work are:

- Modelling of G11 intersection in Eclipse SUMO Software
- Four different configurations depend on vehicle input is used for the same model
- · Traffic plans are subjected to optimisation by using Fixed and Actuated traffic light plans

The research methodology adopted for this thesis was as following:

- 1. Literature Review.
- 2. Selection of the suitable microscopic simulation model from:
  - (a) Available simulation models/software and research carried out in this regard.
  - (b) Selection of the suitable microscopic simulation models based on network flexibility
- 3. Collection of all the relevant data from the field
  - (a) Geometry data (Number of lanes, Lane width and Distance)
- 4. Modelling on the selected microscopic simulation model
- 5. Generation of real world traffic data by using probabilistic distribution
- 6. Collection of results

- 7. Analyzing and evaluating the results
- 8. Comparison of different configurations in the model

Two types of traffic light plans are used for the configurations. Following are the configurations

- · Configuration with vehicle input of Cars only
- Configuration having two lanes fixed for heavy traffic
- Configuration with Equally distributed traffic in 5 lanes

The flow chart highlighting the methodology adopted for this study is shown in Figure 1.1.



FIGURE 1.2: General Methodology Flow Chart

#### **1.7** Organization of the Thesis

**Chapter 1: Introduction** Chapter 1 summarizes the whole research work. It starts with an overview of the background and area of research, then presents objectives of this research study, a brief description of the methodology opted to carry out this research work is given and at the end contribution of this research study made to the scientific literature are penned down.

**Chapter 2: Literature Review** Chapter 2 presents an in-depth review of the related literature. It starts with an overview of the Traffic light control system and then, it discusses the already used traffic light control methods. The core component of this research work i.e. Traffic problem in an intersection in city Islamabad have been studied in detail and the previous methods have been used to control traffic in Islamabad in depth both theoretically and experimentally. Similarly, the concept of traffic control and its implementation were also discussed.

**Chapter 3: Problem Formulation** Chapter 3 gives a detailed overview of the computational setup of intersection in SUMO Eclipse. Various stages involved to carry out a complete study are comprehensively discussed which includes modelling of the intersection, generation of the vehicular routes, selection of appropriate traffic control method, the input of vehicular parameters and choosing the appropriate time step size for the simulation.

**Chapter 4: Results and Discussion** Chapter 4 focuses on the explanation of the results for both fixed and actuated traffic light control in a very comprehensive manner. In the subsequent part of the chapter, visualization of simulation data is presented. Similarly, results of all configuration with cars, equally distributed vehicles and 2 lanes fixed for heavy vehicles were also presented for both traffic light control methods. Results of all configurations with fixed traffic light control have been compared with different lanes and then with actuated traffic control.

**Chapter 5: Conclusion and Future Recommendations** Chapter 5 concludes the findings of this research study and presents possible future directions to extend this work.

### **Chapter 2**

# **Literature Review**

In this chapter, an intensive amount of effort was made to explore every theoretical aspect of this research work. To meet this milestone a detailed literature survey was done on Traffic light control systems. A comprehensive study was done to understand the traffic condition in Islamabad city, the challenges associated. The core component of this research work i.e. Traffic light control methods have been studied in detail and the previous work done on traffic control have been studied in depth both theoretically and experimentally.

The amount of traffic on the roadways is constantly increasing, and the highway infrastructure is carrying more vehicle miles than ever before. The pressure on the transportation system is increasing faster than efficiency can be improved. To prevent this, transportation planners, engineers, and analysts are focusing more on increasing the operating efficiency of highway networks. This necessitates identifying difficulty areas and bottlenecks in a system that require capacity enhancements, as well as developing strategies that address the system's demands, which are frequently constrained by social, environmental, and economic considerations. Modeling traffic operations with computers has unrivalled advantages, allowing traffic engineers to provide acceptable solutions for addressing a particular problem at a cheap cost and with minimal work. As a result, traffic simulation models are becoming increasingly common in transportation facility development and operational analysis.

Traffic light control systems have been extensively researched and implemented in operation. To forecast vehicle density at an intersection, physical sensors and devices such as loop detectors and video cameras bearing content-analysis capabilities) can be used to detect and identify vehicles [65] [27].

#### 2.1 Simulation

Oxford Dictionary defines simulation as the technique of "imitating the behavior of some situation or process" by means of "a suitably analogous situation or apparatus".

[46] describes simulation as "a numerical approach for doing computer experiments that may have stochastic features, be microscopic or macroscopic in nature, and use mathematical models that explain the behaviour of a transportation system over long periods of time." Computer simulation can be defined as the process of:

- Designing a computerized model of a system
- · Conducting experiments with this model
- Understanding behavior of system of evaluating various strategies for the operation of the system

#### 2.1.1 WHY SIMULATE?

Traffic simulations has shown valuable analysis and evaluation tool for evaluating transportation systems. It can be useful in many areas like transportation system designing, traffic operations, and evaluation of management alternatives. Simulation's advantages includes cost effectiveness, risk-free nature, and high-speed benefits. Figure 2.1 shows applications and benefits of microscopic simulation models.



FIGURE 2.1: Application Area and Benefits of Microscopic Traffic Simulation Models [55]

It is not always possible to test transportation projects in the field. Rather, It's not always doable. In such conditions simulation models are great advantage. Field testing is more costly, less safe, and less versatile than simulation. More over, it also provides an objective framework in which to evaluate different alternatives. Advantages of simulation modeling cited by [12] and [16] are as following:

- field implementation are much expensive than simulations. In addition, Simulating traffic operations eliminates the disruptions that typically accompany field studies.
- · Other analytical approaches may not be appropriate
- Many factors that cannot be controlled in the field can be maintained constant in simulation modelling. This may reveal which factors are most significant and how they interact
- Experiment can be done offline instead of time taking trail and error approach
- · Can use future parameters to do predictions
- Time limits can be set according to time frame available
- Can handle interacting queuing processes
- · Traffic demand can be varied according to time and space

#### 2.1.2 TYPES OF TRAFFIC SIMULATION

Traffic simulation is the "Process of applying simulation for traffic applications" Depending on purpose of modeling traffic; it can be "Macroscopic", "Mesoscopic" or "Microscopic" models. The simulation model chosen should be appropriate for the intended application. This decision is usually based on a compromise between the model's accuracy and precision and the development expenses, data requirements, and simulation time.

#### 2.1.3 Macroscopic (macro) models

Individual vehicle movements are not included in macroscopic models; instead, aggregate representations of traffic flow are used. As a result, they consume considerably fewer computer resources, yet they are unable to predict vehicle-level readings accurately. A deterministic macroscopic simulation is important for network design and optimization because of its deterministic character. Macroscopic models, on the other hand, can measure speed, flow, and density.

#### 2.1.4 Microscopic (micro) models

Microscopic models capture the behavior of vehicles and drivers in much more detail and are capable to capture individual vehicle movements. Traffic is simulated using car-following relationships, lane-changing, and other processing logic, which describes the vehicle behavior. This technique is useful for a wide range of applications but requires more computational resources. Random number generators are involved and calibration of these models requires more effort, and are difficult to optimize model parameters, e.g. signal settings.

#### 2.1.5 Mesoscopic (meso) models

Mesoscopic models include features from macroscopic and microscopic models. Usually, results from microscopic models are aggregated for Results from microscopic models are usually combined and used in mesoscopic models. This cuts down on the amount of time it takes to run the simulation. Mesoscopic models have a lower degree of resolution than macroscopic models when it comes to tracking individual cars. They don't represent lane change or automobile following behaviour, for example, whereas microscopic models do. Hybrid Simulation is a term used to describe this method.

#### 2.2 Traffic Signal

Traffic congestion was a part of city life as far back as Roman times. Roads were built out in such a manner to carry traffic from all directions to a single centre point which leads to bad city planning, which was a major problem back then, as it is today. Julius Caesar restricted wheeled traffic from Rome during the daytime in the first century BC, a measure that was eventually applied to cities in the provinces. Emperor Hadrian was pushed to reduce the overall numbers of carts allowed into Rome late in the first century AD.

Congestion was serious enough in European cities in 17<sup>th</sup> century that ordinances banning parking on some streets and enforcing one-way traffic were imposed. While the railroad provided immediate relief from the increasing issue of road traffic control, it also caused congestion at city terminals. With its exponential rise in speed and number of cart's transportation, the invention of cars was one of the characteristic problems of urban industrialized society in the twentieth century [49].

The innovations of traffic signal control were conducted by the British Admiralty in the fields of communications and maritime navigation.

#### 2.2.1 Communications

In the late 18<sup>th</sup> century, a chain of optical telegraph stations were built to allow the Admiralty in London to communicate quickly with the naval ports along the southern coast of England. General Pasley's work in the early nineteenth century improved the telegraph's operation by observing the device perfected in France by Claude Chappe, which led to the UK's introduction of the semaphore type of telegraph in 1816. In response to rising accident rates, General Pasley later became Inspector General of Railways, and it was during this period that he started using the semaphore signal to improve contact with locomotive drivers. The London and Croydon Railway's Charles Gregory built the first of these at New Cross in 1842 [4].

#### 2.2.2 Maritime Navigation

With the onset of steamships around the mid-nineteenth century, there was a significant rise in at-sea accidents, resulting in the loss of many ships. A variety of studies were conducted following the work of a Parliamentary Select Committee, which first looked at the subject in 1831. One of the studies used coloured lights to make way for a ship, allowing it to be more visible to other vessels after dark. Oil lamps particularly with clear, red, and green lenses were found to be the most easily seen from a distance, with the least chance of misinterpretation. The study concluded that for navigation sidelights on boats, red and green lights should be used, a proposal that was unanimously accepted in 1858.

These two studies culminated in two collections of inventions, all of which were cuttingedge innovations at the time.

- The widespread use of semaphore signals to monitor railroad traffic
- · Red and green lights are used as visual warning signals

#### 2.2.3 Goal of Traffic Signal Control

By regulating time in space, traffic lights can improve safety at junctions by preventing traffic conflicts between different vehicles and pedestrian movements. People must wait for the green light in order to prevent dangerous confrontations and optimum control. In the meanwhile, they lose time, which causes annoyance, and cars burn more fuel and emit more emissions.

The goals of signalized intersection change with time and according to the traffic enforcement authority's current policy.

The traffic light control as we know it now has its origins in the manual operation of police signalmen. The first automated traffic light appeared around the turn of the twentieth century. The objective of the traffic light at the time was to promote safety by minimising collisions between cars and pedestrians at crossings by providing alternate right of way to opposing movements using an uniform colour code. The influence of traffic lights in terms of network efficiency, given the same degree of safety, has been realised by the research community throughout the years as traffic demand and congestion have increased. So, in addition to grouping conflicting movements at various periods to ensure the safe operation of an intersection, a method to split green time and movements grouping is required to get the best results.

To summarise, the fundamental goal of traffic control is to ensure road safety, which cannot be overlooked or overlooked in any situation. However, if an effective and appropriate control approach is used, signal timing has the potential to increase mobility while also contributing to environmental advantages.

#### 2.2.4 Highway signals

London's traffic congestion was getting worse in the mid-nineteenth century. On December 10, 1868, in front of London's Parliament, the first traffic signal with gas lightening was mounted. Gas signal was handled manually by a police officer using semaphore guns, as recommended by Rail transportation expert J.P. Knight to manage horse carriage traffic. Gas-lit red and green lights were used at night, but they were always changed by a cop. The lights caused a safety risk since they occasionally burst, injuring wardens. Later on in 1912, at other intersection in Paris, a traffic control system with a rotating four-sided metal box placed atop of a glass showcase with the words "Stop" in red and "Go" in white was installed on top of a tower.

In the same year in 1912, Lester Wire, an American police officer, invented the first electric traffic light as vehicle traffic grew. It was first built on August 5, 1914, at the intersections in Cleveland, Ohio. Later on, In Salt Lake City, the first integrated traffic signal system was built, with six connected intersections operated at the same time by a manual switch. A few years later in 1920, the first tetra directional and three-coloured traffic lights were invented by William Potts, a Detroit police officer. He used yellow lights to show that the light would soon change. The city of Detroit was the first to use four-way and three-coloured traffic signals and in the same year, The Acme Traffic Signal Co. install five signals on Broadway in Los Angeles. The signals consisted of semaphore arms with "Stop" and "Go" messages.

Garrett Morgan received a copyright for his design of an electric traffic light in 1923. After witnessing a tragic tragedy, the African American inventor invented his autonomous traffic signal system.For his invention, GE charged him \$40,000.

Experiments in the use of vehicle actuation to make traffic signals open to vehicles using a junction were carried out in the 1930s. An early attempt at this included placing a microphone along the side of the road and having the lights respond to vehicles honking their horns (left). Even though the scheme operated, it was extremely controversial among the locals. Later experiments used electrical pressure mats and pneumatic tubes, which were widely used until the 1970s when they were replaced for permanent installations with inductive loops. Pneumatic tubes, however, are still used for temporary count sites where they are attached to the carriageways floor.

#### 2.3 Traffic Control Settings Overview

The design and timings of traffic signals are among the settings for traffic control. In order to design suitable signal groups, phase composition, and phase sequencing for traffic signal control plans, the structural design of the junction must be specified. Aside from defining the strategy for green time split, there is a requirement to set bounds for timing variables when it comes to traffic signal timing. The geometric arrangement, traffic vehicle characteristics, and driver conduct all have an impact on these factors. Three colours are used for road traffic and two colours are used for pedestrian traffic. The order and meaning of the colours varies significantly between nations, but the overall goal remains the same and is constant within the country. The traffic controller has three durations: permissive, change, and clearance, which are represented by the colours green, yellowish /flashing green, and red for all traffic lights, accordingly.

#### 2.3.1 Design of Signal Plan

The traffic flow in the junction, or more precisely, in the stop line, is controlled by traffic signal control. During recurrent periods, permission is provided for traffic to cross from one or more lanes (or traffic streams) of one intersection approach. A phase is a period of time during which a certain set of lanes (or traffic streams) gets access to the intersection.

The definition of the term "phase" requires more clarification. In an examination of the literature on traffic control, the words "stage" and "phase" caused considerable confusion because they are used interchangeably. The term "phase" is used in the United States and Australia to represent the concepts of "stage" and "reverse." Furthermore, to add to the confusion, several nations employ a similar term (for example, "fase" in Portuguese) with similar spelling, increasing the chances of misinterpretation. Another argument stems from the fact that under simple traffic control, the two phrases can be interchanged without repercussions because they can have the same meaning in some situations. The right usage of both words is critical in more complicated traffic control when the number of phases does not equal the number of signal groups (stages). To simplify and avoid misunderstanding, this paper uses American language, with the term "stage" being substituted with "signal group."

#### 2.3.2 Timing of Signal Plan

Green time (minimum and maximum for actuated phases), Yellow time, All Red time, Vehicle Extension for actuated cycles, Green time for pedestrian and Flashing green time for pedestrian phase, and Cycle length must all be specified as part of the signal design timing.

#### **Green Interval**:

The green time, also known as the permissible period, signifies that a traffic stream is permitted to enter the junction. The minimal green duration should be sufficient for drivers to react to the green light and satisfy their expectations. If the minimum green duration is too lengthy, it might cause delays at the junction, and if it is too short, it can disappoint drivers' expectations and compromise pedestrian safety.

Only the pedestrian reaction time and time to leave the curb can be included in the minimum value for green time for pedestrians, after which the clearing time can be utilised to cross. Local agency policy generally determines the length of the walk interval.[35].

Method	Value (s)	Description
TRB (2010)	Major-street: 10 s Minor-street: 8 s Left-turn movement: 6 s	Vehicle signal group
	Major-street: 7s to 15s Minor-street: 4s to 10s Collector, Local: 2s to 10s	Minimum green time to satisfy driver expectancy
Koonce et al.	Distance between sensor and stop-line:	
(2008)	$0$ to $7.5m \rightarrow 5s$ $23$ to $30m \rightarrow 11s$ $7.5$ to $15m \rightarrow 7s$ $30$ to $38m \rightarrow 13s$ $15$ to $23m \rightarrow 9s$ $38$ to $46m \rightarrow 15s$	Minimum green time to satisfy queue clearance can be dependent of sensor location to stop-line
Manual of Uniform Traffic Control Devices, MUCTCD (2003)	≥ 7s; ≤15s 4s, if pedestrian volumes are low	Pedestrian
Pavel (1974) Vehicle signal group: 5s-10s Pedestrian signal group: 5s		Values recommended in Germany (Guberinic et al., 2007)

FIGURE 2.2: Green interval

#### Yellow Interval:

Yellow time, also known as change interval, indicates that the permission to enter the junction is about to expire through the yellow colour, giving drivers enough time to perceive and react to traffic light colour changes, as well as the distance required to safely stop or travel through the intersection. As shown in Table 1, there are a variety of suggestions for setting the yellow time..

#### All red Interval:

The period between the end of the yellow and the start of the green of the opposite signal group is known as all red time. A signalised junction's all-red phase is defined for safety reasons, allowing cars that cross the stop line at the conclusion of the yellow phase to exit the crossing before the following phase. The clearing time is thus determined by the junction

Method	Value (s)	Description
McGee et al. (2012)	$t_{yellow} = t_r + \frac{1.4\nu}{2a_{dec} + 64.4g}$	$t_r = 1s$ $a_{dec} = 3.0 \text{ m/s}^2$
Tarnoff (2004)	$t_{yellow}$ =3s , v < 36 km/h $t_{yellow}$ =4s, 36km/h < v < 55 km/h $t_{yellow}$ =5s, 55 km/h < v	Based only on speed.
Manual of Uniform Traffic Control Devices, MUCTCD (2003)	t <sub>yellow</sub> [3; 6] s	Longer intervals for high-speed approaches.
Pires da Costa (1987)	$t_{yellow} = t_r + \frac{v}{2a_{dec}}$ $t_{yellow} = 1 + 0.19v_{cruise}$	$t_r = 1s$ $a_{dec} = 3.0 \text{ m/s}^2$
Transportation and Traffic Engineering Handbook, ITE (1982)	$t_{yellow} = t_r + \frac{v}{2 a_{dec} + 2gG}$	Sum of driver reaction time and deceleration time. $t_r = 1s$ $a_{dec}$ [3.0; 4.6] m/s <sup>2</sup>
"Rule-of-thumb method"	$t_{yellow} = v/10$	Based only on speed.

Note:  $t_r$  = reaction time (s), v = approach speed (m/s),  $v_{cruise}$  = speed before intersection (m/s),  $a_{dec}$  = average deceleration (m/s<sup>2</sup>), g = gravity acceleration (m/s<sup>2</sup>), G = grade of approach (%)



layout, road speed, and municipal laws. As shown in Table, there are a variety of recommendations for how to establish the all-red period.

#### 2.4 Different techniques used for traffic signal control

Many traffic management systems have already been developed in the real world. They've used a variety of methods to manage traffic flow. These methods are listed further down.

#### 2.4.1 Manual traffic control management

This is the most basic type of traffic management, and it relies heavily on human intervention. In this system, a traffic police officer stands at each cross-section of the road and uses a signboard to monitor traffic flow. If there is a high volume of traffic on the lane, the traffic police will send the vehicle driver a signal to move or not. He may also identify an emergency vehicle on the road and assign priority to the lane in which the emergency vehicle is travelling [28]. However, in the event of several cases, she or he will become disoriented and unable to control traffic flow. This process is more effective than any other method, but it is inefficient since it involves humans as part of the system. In several cities, the lack of qualified traffic police officers exacerbates the problem. Since traffic vehicle pressure does not equalise across all

Method	Value (s)	Description
McGee et al. (2012)	$t_{clearance,allred} = \frac{W+L}{1.47v} - 1$	
Manual of Uniform Traffic Control Devices, MUCTCD (2003)	≤ 6 s	Reduction of 1s to account the delay of reacting to green signal by conflicting movement
Traffic Control Devices Handbook (ITE) (2001)	$t_{clearance,allred} = \frac{P+L}{v} - 1$	Reduction of 1s to account the delay of reacting to green signal by conflicting movement
Pires da Costa (1987)	$t_{clearance,allred} = \frac{d_t + L}{v_t} - \frac{d_a}{v_a}$	t is vehicle leaving of intersection a is vehicle entering at intersection
	$t_{clearance,allred} = \frac{W+L}{v}$	When there is no pedestrian traffic. Time to place the vehicle outside the area of conflict
Transportation and Traffic Engineering Handbook, ITE (1982)	$t_{clearance,allred} = \frac{P}{v}$	Used when there is low pedestrian traffic. Time to place the vehicle at a point directly in front of pedestrians waiting to use the crosswalk.
	$t_{clearance, allred} = \frac{P+L}{v}$	Significant pedestrian traffic. Time for the vehicle to clear both the cross street and the pedestrian crosswalks.

Note: W= width of stop line to far side no-conflict point along the actual vehicle path (m); L = length of vehicle (m); P = width of the stop line to the far side of the farthest conflicting pedestrian crosswalk along the actual vehicle path (m); d = distance between stop line and conflict point of vehicle i (m);

roads at the same time, a traffic light should be operated that detects the traffic conditions and adjusts traffic plan correspondingly. The efficacy of the system is determined by the person's expertise and ability [8].

#### 2.4.2 Automatic Traffic Management Technique

An electronic traffic management method is recommended to remove the majority of the flaws in the manual traffic control system. Simple three-colour traffic signals are used in this system: Red, Green and Amber. The green light is normally for 120 seconds at each lane [63], but in certain parts of the city where there is less traffic, the duration of green timing is less than 120 seconds. It is entirely dependent on the density of traffic in specific area. Amber light glows for 20 seconds before the green light; signalling that you should accelerate. The red light is illuminated at all times, signalling that all vehicles must come to a complete half. This system is unable to identify the emergency vehicles such as ambulances, VIP cars, and other similar vehicles. It is uniform in the treatment of all vehicles.

Since red and green duration have been set, and both signals change in order and for midnight, all red and green signals are manually turned off and amber light id turned on. As a result, there is a chance that emergency services will be delayed at peak hours. As a result, this approach can be ineffective at times.

#### 2.4.3 Image Processing based Intelligent Traffic Management Technique

Cameras are used in this method to cater images of the traffic volume on the lane. These are mounted on a tall pole so that they can cover a larger area. A computer chip examines the camera's picture to detect cars on the road [63] which is used to analyse traffic density and transmit this data to traffic signals. Then, traffic plans will automatically get update based on this information. This method isn't always suitable because the camera can't cover long stretches in heavy traffic, and the picture taken by the camera isn't always visible in heavy rain.

#### 2.4.4 Traffic Management system using Wireless Technologies

The emergency vehicle is fitted with an RF transmitter, and the RF receiver is placed on the signal pole in this procedure. When an emergency car approaches the junction, it sends a signal to the RF receiver and then sends a signal to the main control system. The control system will then measure the estimated time for the green signal when an emergency car is going while leaving the red light to the remaining lane. The car would then be able to pass with ease [53]. This method effectively manages traffic congestion while also providing options for emergency vehicles. It collects vehicle density on the road using sensors, load cells, and other devices.

#### 2.4.5 IRIS (Intelligent Roadway Information System)

The Minnesota Department of Transportation created an open-source Advanced Traffic Management System (ATMS) software initiative. Transportation authorities use it to keep track and control traffic congestion. Previously, the system lacked traffic statistics and was primarily used to monitor traffic congestion. However, IRIS can offer actual information on traffic to detect traffic densities, regulate traffic flow, and broadcast traveller information [8]. Advanced Traffic management System (ATMS) functionality is used by IRIS under GPL licence. ATM aids in the reduction of time, the enhancement of highway capacity, and the provision of safer travel directions in general. ATMs is made up of a number of expensive tech implementations. The costs of routine repairs have also been increasing.

#### 2.5 Intelligent Transportation Systems

Intelligent Transportation system (ITSs) have been developed to reduce the amount of time spent or making decisions when driving by offering creative services such as high-tech traffic controllers and various modes of transportation. TLC includes a 4-way traffic intersection to simulate traffic light control. In a four-leg junction, the simulation involves a junction of four different entrances and 4 distinct exits. Furthermore, all roads in the simulation have three lanes. For each strip, two detectors are positioned in the road.

A signalised junction is designed to separate competing traffic movements in time instead of space, allowing them to proceed effectively and safely. The signal group, a series of lights that corresponds to a specific movement in the intersection, is the fundamental unit of a traffic signal control system. During the phase's green-time, a phase is a collection of signal groups that relate to a specific set of movements that have the right of way. These movements have been selected so that they can run in parallel without interfering with other traffic flows.

when each phase of a signal loop has been served once, it is called cycle time (also known as cycle length). Signal control algorithms are typically designed to optimise the traffic flow using the phasing scheme, the split, and the offset settings. The right-of-way signal groups and their subsequent order in the cycle are determined by the phasing scheme. The green-time allocated to each phase is represented by a break, which is a distribution of the cycle time to the different phases. An offset is a difference in time between the start of a new cycle and a reference point in time. Offsets have been effectively used to achieve success along a corridor by generating a "green wave" for vehicles travelling in one direction or reducing the number of stops and waiting times at intersections along the corridor.

Traffic signal control (TSC) has been optimised using a variety of methods, according to researchers. Several high-quality reviews have previously been published on the subject of TSC settings [36] [9]. Fixed-time controllers, actuated controllers and adaptive controllers are three major types of traffic controllers.
**Pre-timed:** During a loop, the traffic signal gives each approach a set duration of green time. This green cycle is set for each interval for a certain amount of time, whether it's an hour, a few days, or forever.

**Actuated:** During a loop, the traffic signal gives each approach a certain amount of green time. This duration can be increased depending on the number of vehicles arriving at the green-lit approach as detected by a detection system. A maximum green time definition also limits the length of each green interval.

**Adaptive:** The traffic signal gives each intersection approach green time dependent on expected arrivals for a loop. The amount of green time given to each solution varies as arrivals shift from cycle to cycle.

## 2.6 Fixed time Control (Pre-Timed Control)

Based on historical traffic data, an off-line fixed signal timing plan is developed. Just once, off-line, are the loop length and splits optimised based on historical traffic counts, and then deployed. These systems do not run in real-time and are only effective when the traffic flow at the intersection is steady and consistent during the day. However, a variety of factors, such as accidents or road maintenance, may cause traffic to change. As a consequence, a predetermined control system is unable to respond in real-time to traffic demand. Following that, the length and order of all green phases remain constant and do not respond to traffic demand fluctuations. Fixed-time controllers do not need loop detectors at intersection approaches to operate, so they are much less expensive to implement than actuated or adaptive controllers.

Fixed-time Traffic signal control techniques are best suited for traffic signals with a relatively steady and predictable flow of traffic. These strategies use offline optimization algorithms for TST and end up with a preset cycle length and split setting plan based on previously observed traffic data. The main aim of this strategy is to accomplish a broad goal, such as minimising average delay or increasing network power.[73] [50] built the first models, which laid the groundwork for fixed-time traffic management techniques by reducing average delays. Since the traffic system in cities is extremely complex, any minor disruption, such as a traffic accident or construction activity, may drastically alter traffic volumes and make a predetermined traffic signal plan ineffective.

The traffic light controller optimises phase timings based on real-time inputs from loop detectors mounted near or immediately before the intersection's stop line using this method [53] [60]. The presence of loop detectors is detected, and the number of vehicles passing over them is counted. The light controllers changes travel intervals based on the vehicle density at an intersection using data obtained in real-time.

#### 2.6.1 Webster Method

Webster suggested mathematical model for determining the optimal cycle length to reduce vehicle delay. Other researchers discovered that when the maximum flow ratio is close to unity, this model fails. As demand is roughly equal to lane availability, this occurs[64]. Webster created a stochastic delay model to estimate the total vehicle delay. Webster calculated the optimum cycle length based on this model to reduce overall vehicle lag. Other researchers have found that when the maximum flow ratio is close to unity, this model fails. As demand is roughly equal to lane availability, this occurs [79].

Webster's method necessitates the calculation of vehicle density 'q' for all movements on all paths to the junction, as well as the saturation flows 's' of the approaches. The flow ratio is calculated as the ratio of demand of traffic flow rate to the saturation flow rate of specific lane groups. Saturation flows are calculated based on lane width [73] [72]. These variables are utilized to calculate signal cycle length, which is the amount of time it takes to finish a signal series. The cycle period contains both phases' missed cycles as well as the green time.

Since drivers without the right of entry must wait for the green signal, traffic signals cause delays [61] [51]. Long delays at signalised intersections, according to [76] and [30], facilitate red-light running as drivers become impatient. As a consequence, the calculation of delays and queue lengths that may occur as a result of the implementation of a strategy for signal control is sacred and forms an integral part of the design of traffic signals. Since delay is the additional travel time encountered by drivers, passengers, or pedestrians in addition to the actual travel time, it is essential to transportation management agencies for policy and decision making [58]. It's a crucial factor in calculating junction productivity and network capacity [25].

For the study and design of road junctions for optimal control systems, advanced societies have developed highly sophisticated tools that they use instead of mathematical models. Surprisingly, the majority of today's junction analysis and design software has either a direct or an indirect connection to Webster's analytical models, which were developed in 1958. The Split Cycle Offset Optimization Technique (SCOOT) and SPOT, a decentralised, social as well as economic optimization method for real-time signal settings real-time specifically for separated junctions based on arrival flow pattern predictions, are two examples of software. SCOOT and SPOT are currently being used in the United Kingdom and Italy, respectively; other countries have also implemented them[54] [45].

## 2.7 Actuated Time Control

Information collected through actuated control, the process of detecting the presence of a vehicle through sensors in a traffic signal system, is used to adapt the control operation based on any fluctuations in traffic demand. It helps decide the time for which a green signal operates depending on traffic demand at the time. This value can differ from cycle to cycle since the cycle length and green time allocation is unknown in advance. The parameters on which this process is based, for each signal group, are as follows:

- Maximum green time
- Minimum green time
- Vehicle extension

During the minimum green time, traffic signals are green independent of the demand of traffic. After a minimum green time, actuated control uses the gap-seek logic to respond to traffic fluctuations, the green signal is extended until the presence of vehicles is detected or until the maximum green time has been reached [78].

For intersections almost in saturated conditions, actuated control behaves like a fixed-time control intersection, with signal groups assuming the maximum green time value permitted by the traffic signal control timing.



FIGURE 2.5: Green time extension

There are also others features regulation options for actuated control, namely phases can be skipped, if there is no call in the sensor, thus allowing the controller to reallocate the unused time to a subsequent phase. Another approach is incorporating decision-making, where a timing plan is selected from a library according to actual traffic conditions collected through sensors in intersection.

When timed to prioritise vehicle movements only, actuated signals prioritise movement along the primary corridor and can create barriers for cross traffic and pedestrians. Where actuated signals are used, they should be timed to be as sensitive to activation as possible, with as little delay as possible. Actuated traffic controllers are designed to assess real-time traffic conditions using vehicle detection and respond appropriately to preserve the highest possible level of efficiency under varying traffic conditions [62].

Modern traffic controllers, on the other hand, are advanced electronic timers that are programmed to respond in a specific way when a sensor's switch (or detector) is actuated. Beyond the fact that a service request occurs, the traffic controller has no idea what the actual state of the traffic is, and it has no specified aim to accomplish by objectively changing its performance. The output is solely controlled by the stimulus-response relationship, with little regard for the intersection as a whole.

#### 2.7.1 Traffic Actuated controller

Inductive detectors are used in traffic-actuated monitoring methods to observe the current traffic condition. During the red process, the traffic-actuated controller must be able to decide if the last vehicle in the queue created at the stop line has left [81]. This detection is done

by calculating the distance between vehicles and is useful for extending or terminating green time efficiently. If the distance between the vehicles exceeds the threshold limit gap, the green time is over [5]. The efficiency of the actuated system is influenced by the optimum location of detectors at an intersection.

#### 2.7.2 Traffic Simulator

Simulations are done using traffic simulator SUMO. Simulation of Urban Mobility SUMO [2] is an open-source road simulation package, developed by the German Aerospace Center (DLR), that simulates traffic behaviour, allows for road construction, traffic light policy implementation, and traffic data collection. In comparison to other simulation tools, SUMO [1] offers more model expansion options with less complexity. Furthermore, Sumo allows users to simulate traffic using geographic data.

SUMO has a tool ActivityGen that allows to simulate individuals travelling to and from work or school. As a result, extremely comprehensive information such as where inhabitants live and work, as well as the location of schools, is required. Previous studies focus on using geographical population distribution to create traffic demand. Codeca et al. [17] reconstructed a 24 hour scenario of Luxembourg city including geographic population distribution, morning and evening rush hours, as well as buses and bus terminals. However, accuracy cannot be proven without a real-world comparison and correction. Asano et al. [11] used ActivityGen to produce OD data for a scenario in Kobe, Japan and discovered that adjusting the speed limit of the road brought the simulated data closer to the traffic census data using the default parameters.

# Chapter 3

# Methodology

An overview of the whole process is discussed in this chapter, which is used to achieve the desired target. The description of all the major steps that were performed is also shown.

A virtual experiment employing traffic simulation is an efficient technique to address current and future issues. The most critical factors in creating a real-world simulation are a detailed road network and the associated traffic demand. Other factors that influence simulation accuracy include the road choice algorithm [47], simulation software package selection [57], traffic light control methods [43], and so on. Although most of the modelling software packages can import an open-source city map as a road network [57], real-time traffic demands are not easy to cater.

First, we modelled the desired intersection by using Eclipse SUMO which is a microscopic traffic simulator [44]. There are six major steps in the signal design process. They contain the following:

- · Designing of Phase
- Calculation of yellow time and clearance time
- Persistence of cycle length
- Assigning of green time
- The performance evaluation of the above design

It is difficult to model traffic flow directly from the microscopic information that controls individual vehicle motion. Microscopic traffic flow modelling aims to explain the dynamics of individual vehicles as a function of its neighbouring vehicle's positions and speeds [29]. Certain characteristics of traffic flow, on the other hand, are independent of the details of the underlying processes. To avoid collision between vehicles requires the average velocity V in a locally uniform traffic flow to decrease with an increase in vehicle density  $\rho$ . To find the averages, some relations between V and  $\rho$  have to be known.

$$\frac{dV(\rho)}{d\rho} < 0 \tag{3.1}$$

V reaches a finite value  $v_{max}$ , for very low traffic volumes, while no motion is possible at the maximum density  $\rho_{max}$ , when every vehicle stands bumper to bumper.

$$V(0) = v_{max} \tag{3.2}$$

$$V(\rho_{max}) = 0 \tag{3.3}$$

The resulting average flux of the vehicles which is defined as the average number of vehicles passing a given cross section per unit time  $Q(\rho) = \rho V(\rho)$ , is a function that disappears at  $\rho = 0$  and  $\rho = \rho_{max}$  and has a maximum at a certain intermediate density.

#### 3.1 Model Formulation

Phase design aims to divide opposing movements in a junction into different phases so that movements within each phase are not in collision. A large number of steps are needed to isolate all of the movements without causing conflicts. We have considered a four-legged junction with through traffic and right turns to demonstrate the different phase plan choices. Moreover, turning left isn't taken.



FIGURE 3.1: Description of Phases on an intersection

# 3.1.1 Four Phase Signals

This phase's plan is suitable for urban areas where turning movements are comparable to through movements where through and turning traffic must share the same lane. When turning movements are poor, this plan could be inefficient. A third step is formed by the nonconflicting right turn flows of 7 and 8.



FIGURE 3.2: Detailed Description of Phases on an intersection

#### 3.1.2 Roads Network

The intersection is comprised of 4 roads with 5 lanes. A lane includes geometry, knowledge about the vehicle classes allowed on it, and highest speed that can be reached. Hence, more than two edges are used to represent lanes of the roads [40]. SUMO provides tool that allows to communicate with Python program. In order to simulate the traffic behavior, SUMO requires the simulation files that are written in xml format containing configurations regarding to simulation.

Name	Extension	Description	
Nodo	<name> nod yml</name>	Describes all the node in the	
Noue		simulation network	
Edge	<name> ada yml</name>	Describes all the edges between	
Luge		the nodes	
		Describes number of lanes and	
Type		speed limit of roads	
Network	<name> not yml</name>	Describes the environment	
		(roads and intersections)	
		Describes the route of all vehi-	
Koule		cles that are in network	
Configuration	<name> sumoof a</name>	Describes the configuration of	
Cominguration	<iname>.sumocig</iname>	simulator	

TABLE 3.1: Simulation files description

Network, route and configuration files are essential for simulation. Node, edge and type files are required to construct the network file. SUMO provides some additional packages that allow user to edit every element of road infrastructure. Among these packages provided by sumo, following are used in this thesis research.

Road network can be created with the help of a tool "netgen" or "netconvert". Other popular formats, such as shapefiles and Open Street Maps, are also supported. We have used Open Street maps with network files for model formulation. These applications import digital road network from different sources for example: google map or open street maps and generate road networks that can be used by other tools from the package [39].

#### 3.1.3 Netconvert

It is command line tool that come along with SUMO software. It is used to convert the node, edge and type files into network file. It is also used for the conversion of map file from

other sources like OpenStreetMap (OSM) and VISUM-network. In the case of singe agent, the network is constructed from node and edge files while in the case of multi-agent, network is constructed from OSM file.

#### 3.1.4 Polyconvert

It is command line tool used to import geometrical shapes (polygons or points of interest) from different sources. It is used to improve the visual experience of simulation. The map of multi-agent environment is improved by polyconvert tool. The Figure 3.3 compare the basic and improved network of multi-agent environment.



FIGURE 3.3: Improved network

#### 3.1.5 NetEdit

NetEdit is GUI based network editor. It is used for editing all aspects of existing network. In this thesis it is used to edit attribute related to intersection and roads. For intersection, it is used to edit name, routing, phase, phase cycle and phase duration and for roads, it is used to edit name, width and incorrect connections. It is the best tool to edit those networks that are imported from other sources like OSM and VISUM-network.

#### 3.1.6 TraCI

TraCI is the short term for Traffic Control Interface. It gives access to simulation values like vehicle position, velocity, acceleration, intersection phase etc. It is imported in Python programming language as a module and gives access to simulation in Python program. By using TraCI useful information like state is obtained from the simulation environment and action is send to simulation environment.

#### 3.1.7 Simulation

SUMO provides two type of simulations, one is terminal based and other is Graphical User Interface (GUI) based. Terminal based simulation works in background and does not show road network and traffic. It is used to retrieve simulation data. On the other hand, GUI based simulation is used for presentation purpose as it shows the road network and traffic on it. Terminal based simulation is much faster than GUI, that is why it is used in training process of reinforcement learning agents.

#### 3.1.8 Population Definition

Traffic data are crucial when simulating real-world traffic. Some people look for the realworld data to develop a model, while others use real-world data to validate their simulation results. The majority of current research uses three ways to obtain real traffic demand: trajectory data from cameras, speed data from induction loops detectors, and floating car data (FCD) from data loggers [70]. It is becoming increasingly difficult to acquire approval to place cameras on public highways due to increasingly severe privacy policies. Installing induction loop detectors has various problems, and paving induction loops on each lane is an additional and frequently impossible amount of research work.

For many researchers, obtaining real traffic data is always a challenge. The majority of research projects involving the creation of traffic scenarios benefit from working with the government to access data sources.

It is important to generate traffic with high reliability. In order to maintain this, different types of distribution are used to mimic traffic Experimental Setup behavior. Three type of distribution are common in use. These are Uniform, Gaussian and Weibull distribution.Uniform distribution is one of the simplest distribution and it generates almost equal number of vehicles at every timestep of simulation. This is not usually the case of real-world traffic generation. So, it is not a good idea to use this distribution to generate traffic in simulation. Gaussian distribution generates different number of vehicles but does not mimic the real-world traffic as it gradually increases and decreases. In real-world, traffic flow is high in peak hours especially in morning and then decreases slowly. Such type of behavior is shown in Weibull distribution and that is why it is used in this thesis research.

Traffic Scenario	Traffic Generation		
Low	600		
High	3000		
East-West(EW)	1500		
North-South (NS)	1500		

TABLE 3.2: Traffic generation in different traffic scenarios

The total number of cars generated in each traffic scenario and timestep at which these cars entered the simulation environment is known and this information is used to define the source and destination of generated cars. Let start with the simplest case of four-way intersection. Each compass direction of intersection has some probability that defines the portion of cars being generated at that side of intersection. This probability distribution of cars in different traffic scenarios are listed in Table 3.3.

Traffic Scenario	East	West	North	South
Low	0.25	0.25	0.25	0.25
High	0.25	0.25	0.25	0.25
East-West (EW)	0.40	0.40	0.10	0.10
North-South (NS)	0.10	0.10	0.40	0.40

TABLE 3.3: Traffic generation in different traffic scenarios

In the case of low and high traffic scenarios, car generation probability is same for all directions (East, West, North and South). The Table 3.3 defines the source of generated car, but destination of car still needed to be defined. In this thesis research, it is decided that 60% of cars goes straight, 20% of cars goes left and the remaining 20% of cars goes right. Table 3.4 lists the exact probability distribution for all possible combination of source and destination for low and high traffic scenario.

East-West and North-South traffic scenarios are generated to mimic the busy roadways in big cities. These scenarios also force the reinforcement learning agent, not to be biased towards one compass direction of intersection. In the case of EW traffic scenario, 80% of the traffic is coming from East and West directions and remaining 20% is coming from North and South directions. The cars generated on East and West directions have the same destination probability that is described in low and high traffic scenarios. In the case of NS traffic scenario 80% of the traffic is coming from North and South directions and remaining 20% is coming from East and West directions. Exact probability distribution for all possible combinations of sources and destinations for East-West traffic scenario is given in Table 3-5 and for North-South traffic scenario, it is given in Table 3-6.

Directions	Source	Destination	Probability
Straight	East	West	0.15
	West	East	0.15
	North	South	0.15
	South	North	0.15
Left	East	South	0.05
	West	North	0.05
	North	East	0.05
	South	West	0.05
Right	East	North	0.05
	West	South	0.05
	North	West	0.05
	South	East	0.05

TABLE 3.4: LOW	<sup>r</sup> and High traffic	scenario probability	y distribution
	0	1 1	

Directions	Source	Destination	Probability
Straight	East	West	0.24
	West	East	0.24
	North	South	0.06
	South	North	0.06
Left	East	South	0.08
	West	North	0.08
	North	East	0.02
	South	West	0.02
Right	East	North	0.08
	West	South	0.08
	North	West	0.02
	South	East	0.02

TABLE 3.5: EW traffic scenario probability distribution

Directions	Source	Destination	Probability
Straight	East	West	0.06
	West	East	0.06
	North	South	0.24
	South	North	0.24
Left	East	South	0.02
	West	North	0.02
	North	East	0.08
	South	West	0.08
Right	East	North	0.02
	West	South	0.02
	North	West	0.08
	South	East	0.08

TABLE 3.6: NS traffic scenario probability distribution

To generate traffic in SUMO simulation environment, route file is required. This file contains all the information about source, destination, vehicle departure time and route Experimental Setup between source and destination for each car. The above discussion describes when the car is generated in network and what is its source and destination location. The missing information is the optimal route between source and destination location. Finding optimal path does not seems to be good in the case of one intersection, SUMO provides package named DUAROUTER (Dynamic User Assignment Router). This package provides pathplaining algorithms (Dijkstras,  $A^*$ , Contraction Hierarchies and CHWrapper) to find the shortest path between source and destination location. In this thesis research  $A^*$  algorithm is used to find the optimal path between source and destination location.

Activity based demand generation (activitygen) generates demand from a description of the population in the network [20]. It is comprised of the network file which is the SUMO .net file, statistics .sat file contains the description of the population, .trips file generates the routes of the vehicles [13]. The statistics file contains the information related to inhabitants, households, children age limit, retirement age limit, car rate, unemployment rate, foot distance limit, coming and outgoing traffic. This file generates the vehicle as per the demand of the population. All these parameters related to Islamabad city are listed in the table given below.

Parameters name	Value type	Value
Inhabitants	Integer	1.015 million
Households	Integer	336,182
Children age limit	Integer	18
Retirement age limit	Integer	60
Car rate	Float [0,1]	0.5
Unemployment rate	Float	4.45
Foot distance limit	Float	350
Incoming traffic	Integer	5000
Outgoing traffic	Integer [0,inhabitants]	5000

TABLE 3.7: Population parameters used to generate vehicular input

Figure 3.4 and 3.5 demonstrate how to import networks from OpenStreetMap (osm) [3] and the flow chart used to import network files combined with OpenStreetMap file respectively. Additionally, netconvert reads the "XML" representation of a road network that offers the readable form when defining a road network.



FIGURE 3.4: Original OpenStreetMap network of Srinagar Highway



FIGURE 3.5: Road Network using Open street map with link of .net files

The goal of microscopic traffic flow modelling is to explain the dynamics of each vehicle as a function of its neighbours' positions and speeds. Car-following and Lane-changing are two dynamic processes that must be considered in general.

We'll start by describing car-following models, assuming a road with a single lane, as a much more sophisticated methods is required when multiple lanes are involved. All cars' positions and velocities are denoted by  $x_i$  and  $v_i$ , respectively, with the index *i* increasing in the downward direction.

#### 3.1.9 Car Following Model

The development of a car-following theory will always begin with the reasonable assumption that any change in the velocity is solely made if the current velocity does not coexist with a desired velocity  $V_{des}$ , determined by safety considerations, legal constraints, and other factors. A relaxation on some time scale d is the simplest dynamic that explains attempts of a driver to approach the desired velocity.

$$\frac{dv_i(t)}{dt} = \frac{v_{des} - v_i}{\pi} \tag{3.4}$$

This basic concept form the basis of almost every car-following theory. This dynamical relationship is usually understood as a stimulus-response–ansatz rather than a relaxation process. The reciprocal of  $\pi$  is commonly referred to as the "sensitivity" in this case.

$$\frac{dv_i(t)}{dt} = \frac{v_{i+1}(t) - v_i(t)}{\pi}$$
(3.5)

Pipes was the first to propose this dynamical equation. He derived this equation through the statement that the space between cars needs to increase in a linear manner with speed and then the resulting relationship is to be differentiated. When one solves this equation, it is easily realizable that steady-state solution, in which every car moves at an identical speed, is always stable, implying that effects due to clustering are not present[41].

Since delay times are known to destabilise initially stable systems, the instability of the the steady-state in this model is not surprising when  $\Delta t/\pi$  is large enough. The thorough analysis reveals that for  $\Delta t/\pi = 1/2$ , the stability limit has been reached.

Although random grouping is modelled in this way, the model's applicability is limited to the fact that the dynamics are unaffected by the vehicle distances. Two problems can arise from this fact. Firstly, the cars collide in this model, and secondly the vehicle dynamics are not density-dependent, so the speed-density relationship cannot be obtained from it.

Gazis, Herman, and Potts solved this problem by introducing a relaxation time (sensitivity), which is dependent on the distance between the cars

$$\frac{dv_i(t)}{dt} = \alpha \frac{v_{i+1} - v_i}{x_{i+1} - x_i} \Big|_{t - \Delta t}$$
(3.6)

The equations of the model can be combined, and because the right-hand side is the derivative of alpha with respect to time  $ln(x_{i+1} - x_i)$ , the density–dependent homogeneous steadystate solution is

$$v \propto \ln(\frac{\rho_{jam}}{\rho})$$
 (3.7)

where  $\rho_{jam}$  represents the density of jam, with a velocity of zero. Greenberg's macroscopic theory is supported by this finding. The flux  $q = \rho v$  disappears for vanishing density, however, since the divergence is logarithmic, we have a qualitatively right fundamental diagram.

Edie suggested that a vehicle's sensitivity is also affected by its momentary velocity. He hypothesised that velocity sensitivity is inversely proportional to spacing, giving rise to the following equation.

$$\frac{dv_i(t)}{dt} = \alpha' v_i(t) \frac{v_{i+1} - v_i}{(x_{i+1} - x_i)^2} \Big|_{t - \Delta t}$$
(3.8)

However, the most interesting aspect of Edie's work was that he noticed a discontinuity in the fundamental diagram, being the first to do so. This contributed to the concept of distinguishing amongst two separate "modes of action," free flow and congested flow, which correspond to two different sensitivity  $\alpha'$  values.

$$\frac{dv_i(t)}{dt} = \alpha' v_i(t)^m \frac{v_{i+1} - v_i}{(x_{i+1} - x_i)^l} \Big|_{t - \Delta t}$$
(3.9)

The exponents l and m are free parameters in this case. Obtaining these exponents from empirical evidence, however, proved difficult. The speed–density–relation of a homogeneous flow can be obtained by directly integrating the dynamical equations. As an example, for l=2

and m = 0, the relationship between speed and density is linear, agreeing with Greenshields' findings.

# 3.2 Traffic light control methods

Two types of traffic signal plans have been used for Traffic light control:

- Fixed traffic light control
- Actuated traffic light control

The traffic light control at the intersection is designed according to the following assumptions and limitations:

- 5000 number of vehicles are routed on each leg at the intersection
- No left turn is allowed at the intersection however separate road is used for left turn only
- The minimum time for the green time in fixed traffic control is 50 seconds and 25 seconds for actuated traffic light control
- The yellow light for both traffic light control is set for 5 seconds during the transition from red light to green light
- The main phases are given equal amounts of green time

The Webster method has been used to calculate the least cumulative delay for vehicles at a signalized intersection. Mostly during the approach of development, the fieldwork consists of determining (i) Saturation flow S per unit time on each approach of the intersection (ii) the normal flow "q" on each approach. Traffic lights are generated with a fixed cycle time of the 90s. Webster's algorithm works by first optimizing cycle duration by minimizing vehicle delay for a given intersection and then allocating efficient green times to the given green period phases accordingly.

#### 3.2.1 The traffic light control Algorithm

The key concept is that each traffic light tries to solve the traffic congestion in its front. He does this by looking through the incoming lanes and measuring the lengths of the jams on these lanes. if traffic develops in one of these lanes, the lanes become green for a long period. Apart from these assumptions, a range of factors keeps the device from fluctuating and evolving too rapidly or too strongly. This is accomplished by extending the size of a green period only if a traffic jam exceeds a threshold. Furthermore, the traffic jam must last for a certain period[37] [75] [22].



FIGURE 3.6: Traffic light control flow chart

Where **tr** and **tg** are proportions for red and green phases, **dlook** represents looking distance, **tdecide** represents decision time interval. **nratio** is calculated as (waiting n - waiting e) / waiting n and **nlimit** is decision threshold.

#### 3.2.2 Fixed traffic light control

Every traffic light was generated using a fixed cycle and a default cycle time of 50 seconds. Programs were generated wit 4 green phases:

- Srinagar\_highway\_ISB\_RWP
- Srinagar\_highway RWP\_ISB
- Service\_road\_G11 \_NUST
- Service \_road \_NUST \_G11

lightgray Phases Names	Duration
Srinagar_highway_ISB _RWP	50 sec
Yellow Time	7 sec
Srinagar_highway RWP_ISB	50 sec
Yellow Time	7 sec
Service_road_G11 _NUST	50 sec
Yellow Time	7 sec
Service _road _NUST _G11	50 sec

TABLE 3.8: Duration plan for Fixed Traffic light control



FIGURE 3.7: Green lines showing fixed Signal plan of Srinagar highway facing Islamabad to Rawalpindi having straight and right turn directions



FIGURE 3.8: Green lines showing fixed Signal plan of Service road facing NUST to G11 having straight and right turn directions

#### 3.2.3 Actuated traffic light control

Actuated traffic control is used whenever continuous traffic is detected and it works by extending traffic phases[24]. Subsequent to detecting a sufficient time gap between consecutive vehicles, it switches to the next phase allowing for improved distribution of green-time amidst phases as well as affecting cycle duration against dynamic traffic conditions. It uses phase attributes minimum duration "minDur" and maximum duration "maxDur" rather than duration to define the acceptable range of time duration for every phase. The detectors are placed which



FIGURE 3.9: Green lines showing Fixed Signal plan of Srinagar highway facing Rawalpindi to Islamabad having straight and right turn directions



FIGURE 3.10: Green lines showing Fixed Signal plan of Service road facing G11 to NUST having straight and right turn directions

determine the lengths of incoming lanes. The detector will be positioned on the predecessor lane at a computed distance if the coming lanes are overly short and if there is a series of unique former lanes. A phase's length must not be longer or shorter than predefined thresholds, in addition to the standard value given at the start. The entire algorithm is described below.

Like fixed light control, the actuated traffic light control has also 4 set of green phases same as mentioned above.

lightgray Phases Names	Duration	Maximum	Minimum
Srinagar_highway_ISB _RWP	50 sec	50 sec	17 sec
Yellow Time	7sec		
Srinagar_highway RWP_ISB	50 sec	50 sec	17 sec
Yellow Time	7sec		
Service_road_G11 _NUST	50 sec	50 sec	17 sec
Yellow Time	7sec		
Service _road _NUST _G11	50 sec	50 sec	17 sec

TABLE 3.9: Duration plan for Actuated Traffic light control



FIGURE 3.11: Green lines showing Actuated Signal plan of Srinagar highway facing Islamabad to Rawalpindi having straight and right turn directions



FIGURE 3.12: Green lines showing Actuated Signal plan of Service road facing NUST to G11 having straight and right turn directions



FIGURE 3.13: Green lines showing Actuated Signal plan of Srinagar highway facing Rawalpindi to Islamabad having straight and right turn directions



FIGURE 3.14: Green lines showing Actuated Signal plan of Service road facing G11 to NUST having straight and right turn directions



FIGURE 3.15: Flow chart of Methodology

# **Chapter 4**

# Results

In a large-scale two-dimensional urban network, particularly in the fourth largest Islamabad-Rawalpindi metropolitan area in the country with a population of about 3.1 million, the greatest challenge is to monitor and organize traffic lights. The results of fixed and actuated traffic control methods, on an intersection of Srinagar Highway at the east-west of Islamabad, are discussed in this chapter. The intersection comprises 4 roads that have 5 lanes and 1 single road for left turn only. The goal of this research is to minimize vehicle staying time at the intersection by reducing the queue length of each side of the junction.

The SUMO simulator is a C++-based open-source highly portable microscopic road traffic simulation package designed to handle large road networks. SUMO can be integrated with Open Street Map (OSM) and can use various road network maps. It is possible to import distinct map scenarios from OSM. OSM is an online world map service that enables the export of maps along with various data formats. To find the optimal routes, we import a city map from the OSM map library and incorporate it into SUMO.

We also implemented SUMO TraCI (Traffic control interface) extension which allows for dynamic control for the traffic lights at runtime [74]. Running as a client, and calling the SUMO simulation tools as a server, TraCI not only allows for modification of traffic light timings while in simulation, but it also allows for information access of individual objects, such as vehicles. This makes it possible to extract the vehicle position and velocity matrices needed to calculate delays and queue lengths.

## 4.1 Fixed-Time control (Pre-Timed control)

Fixed time signal control uses predetermined time intervals that, irrespective of changes in traffic levels, is the same every time the signal cycles. Based on historical knowledge, fixed time control offering the highest green period to the heaviest traffic flow. The duration, regardless of the existence or absence of traffic demand, is replicated over and over. Adjacent intersections run at the same cycle length when functioning as part of a system and have fixed backups.

In general, for reasons of regularity, network organization, predictability, and reducing unwanted delays, fixed-time signals are the standard in urban areas. Less initial and continuous maintenance costs are covered for fixed time traffic signals. For the prediction of traffic demand, pre-defined traffic control methods are most preferable. The most commonly used measure of effectiveness for signalized intersections is a delay. Delay to the traffic is computed by counting the number of vehicles in the queue at fixed intervals of time and multiplying this number by the value of the time interval A delay is most commonly used quality for measure the effectiveness of signalized intersection which is to calculate by finding the number of cars in the specified junction [38].

The current signal control method is fixed control where 1 signal and 8 phases are defined. The phase duration varies from one to fifty seconds. There were 10000 vehicles during the simulation run. In the simulation, the speed deviation is set as 0.1 for vehicles. Three sets of configurations are used to evaluate the behaviour of fixed traffic light control. These are

- Cars only in fixed traffic light control
- Equally Distributed traffic with fixed traffic light control
- 2 lanes fixed for heavy traffic in fixed traffic light control

In SUMO fixed-time signal control is optimized using the Webster method (1958). Parameters used for fixed traffic control are: Saturation Flow S = 1900 veh/lane Green Time = 50 s Yellow Time = 7 s Red time = 150 s Around 10000 vehicles were simulated over a given period of about half an hour during each simulation run, roughly reflecting a typical traffic load during rush hour traffic.On the dominating approaches, the minimal delay for traffic at a junction is determined by setting the green intervals of the phases in accordance to the respective ratios of flow to saturation flow.



FIGURE 4.1: Configuration of cars in fixed traffic light control

This graph shows the simulation with the restriction of cars only in the traffic. The traffic light is selected as a fixed control. The highest value of queue length for fixed traffic light control is 163 m against 1850 s time step and the lowest value is 0 m of queue length at 50, 250, 700, 900 and 1550 s time step. Every phase in the traffic light plan has 50 s of green time.



FIGURE 4.2: Configuration of equally distributed traffic in fixed traffic light control

This graph shows the simulation with all types of vehicles such as cars, trucks, buses, bikes and bicycles. The traffic light is selected as a fixed control. The highest value of queue length for fixed traffic light control is 817 m against 880 and 900 s time step and the lowest value is 0 m of queue length at 50, 160, 180, 290, 310, 440, 550, 680, 810, 1070, 1220 and 1330 s time step. Every phase in the traffic light plan has 50 sec of green time.

This graph shows the simulation with the setting of the last 2 lanes fixed for heavy traffic like bus, trucks and loading vehicles. The traffic light plan is selected as a fixed control. The



FIGURE 4.3: Configuration of 2 lanes fixed for heavy traffic in fixed traffic light control

highest value of queue length for fixed traffic light control is 160 m against 24 and 26 sec time step and the lowest value is 0 m of queue length at 50, 250, 700, 900 and 1550 s time step. Every phase in the traffic light plan with 50 s of green time.

The aforementioned graphs show the queue length is affected when the number of vehicles increases gradually. Traffic congestion is heavy during peak hours in the morning and at night. The peak hours of the morning are usually from 7:00 AM to 9:00 AM when people going to their places of employment and 5:00 PM to 10 PM. By comparing this configuration of cars only, equally distributed traffic and 2 lanes fixed for heavy traffic, only cars configuration showing less vehicular delay by minimizing queue length.

Gradually, results show the optimum value of performance rather than rapid decline variations with time as the congestion increases. During the non-congested hours, performance is almost the same, but it shows the variations when the roads are exposed to traffic congestion. When coming to measure the benefits of using bus lanes for private vehicles in the configuration of 2 lanes fixed for heavy vehicles, the simulation did not prove any benefits at all. The major reason for this is because cars that utilise bus lanes either slow down or are blocked by buses. This would cause congestion and results in an increase in queue lengths.

Also, comparison of queue length has been done for four side of intersection to check vehicle input and vehicular delay. Following graph shows the queue length fluctuations for 5 lanes of all sides.

This graph shows the variations in queuing length for Srinagar\_highway\_ISB\_RWP, Srinagar highway\_RWP\_ISB, Service road\_G11\_NUST and Service road\_NUST\_G11. The Left and right sides of the Srinagar highway show the same behaviour of queuing length as both sides have almost equal daily vehicular input. But for the other two sides that are Service road\_NUST\_G11 and Service road\_G11\_NUST, there are varying queue lengths. Also, results show that there is



FIGURE 4.4: Comparison of all four sides of an intersection with taking different vehicular input

a small amount of traffic coming from NUST to G11 so that it has the smallest queue length from all of the other directions. So, there is no need of whole 50 sec green time for the phase NUST\_G11.

### 4.2 Actuated time traffic control

The amount of green time can be adjusted by an actuated signal controller based on information from loop detectors, for every cycle. Actuated signals are suitable where traffic levels fluctuate greatly throughout the day.

Pre-timing signal control implements optimized plans but they have fixed timing, hence, it is not intended to respond to rapid fluctuations in traffic flow. The old practices of pre-timing traffic signals, while simple and without requiring professional workers, are laborious, time-consuming and repetitive. Besides, it is understood that signal timing plans age with time i.e. many traffic lights operate with timing plans built months and even years ago. On the other side, actuated signal control responds to change in demand trends by applying a green time window (minimum green to maximum green) in contrast to the green time in pre-timed signal control. Also unlike pre-timed signal control, actuated traffic signal control systems are responsive to traffic flow fluctuations[42]. In Actuated controllers where phases can be skipped or shortened, depending on the demand and where the minimum and maximum green times for each phase are considered.

The Actuated controller is based on three timers to control the green phase as discussed below: minimum green, vehicle extension and maximum green.

- **Minimum green**: The green period that must be covered by the chosen process, usually regulated by the minimum green time for a pedestrian crossing to ensure that the selected phase secure pedestrian passage
- Extension time (passage time): The amount of green time which the controller extends when a vehicle is detected during a green interval. During the green interval, the passage time is extended to a vehicle travelling from the extension detector to the stop line. If the minimum green has expired, the passage time only comes into effect. Based on a 40 kph speed and the distance between the spot line detector and the extension detector with a normal 3-5 s range, it is measured
- Maximum Green: Maximum green period for the selected phase. As soon as a conflicting vehicle is identified during which the selected stage is orange, the maximum green period commences [19]

For movements with the highest volume between the minimum and maximum green window, actuated regulation often gives greater priority.



FIGURE 4.5: Configuration of Cars as vehicular input in Actuated traffic light control

Like fixed time traffic control, all three configurations have been done for actuated traffic control. The aforementioned graph depicts the simulation with Cars and showing fluctuation in queue length with time. The maximum value of queue length for actuated traffic light control with Cars is 61 m in the time step of 316 s. likewise, minimum value for same configuration is 0 m for 0-44, 75, 106, 153, 176, 178, 237, 239, 310, 321, 345, 411, 432, 433, 469, 544, 596, 671, 680, 726, 799, 853, 941 and 995 s.

By comparing with Car's configuration of fixed traffic light control, there is a reduction in the queue length of Car's configuration of actuated traffic light control. There is a sudden increase in number of traffic which results in long queue but gradually getting reduced for to fluctuation in time of actuated control [80].



FIGURE 4.6: Configuration of Equally distributed traffic as vehicular input in Actuated traffic light control

For Equally distributed traffic's configuration, the maximum value of queue length is 155m across 786 s time step and the minimum value is zero for many time steps. By comparing this configuration with fixed traffic control, actuated traffic light control shows better performance. The queue lengths are comparatively higher than the only cars configuration but this fluctuation has been seen after some interval.



FIGURE 4.7: Configuration of 2 lanes fixed for heavy traffic as vehicular input in Actuated traffic light control

The queue length is affected when the number of vehicles increases gradually. By comparing the configuration of actuated traffic light control with fixed control, it shows different patterns and fewer values of queue lengths. There is rapid fluctuation after the time step of 500 s. The highest value of Queue length is measured to be 260 m and the minimum is 0 m.

# 4.3 Comparison of all Configurations

By comparing all these configurations, configuration of cars with actuated traffic light control has smallest value of queue length than other two. In the configuration of 2 lanes fixed for traffic light, buses blocked the way for other vehicles which tends to increase the queue length than other configurations. There are also fluctuations in equally distributed traffic, buses and truck would decelerate the speed of cars which results in waiting delays and increment in queue length.



FIGURE 4.8: Comparison of all configuration for actuated traffic light control



FIGURE 4.9: comparison of four phases at an intersection of G11

This graph shows about the variations in queuing length for Srinagar\_highway\_ISB\_RWP, Srinagar\_highway\_RWP\_ISB, Service\_road\_G11\_NUST and Service\_road\_NUST\_G11.

Srinagar\_highway\_ISB\_RWP have highest queue length fluctuations throughout the simulations but unlike fixed traffic signal control, Srinagar\_highway\_RWP\_ISB have decrease in volume of queue length with small amount of fluctuations.

# Chapter 5

# **Conclusions and Recommendations**

This thesis has proposed and evaluated a general modelling procedure for microscopic simulation models for traffic optimization. For this purpose, different simulation platforms were studied and evaluated to select one, which best suites the traffic conditions of Pakistan. After going through the literature and previous research, micro simulation software SUMO was selected for its obvious advantages over the other available micro simulation platforms. The thesis proposed a flow chart to conduct any micro simulation study.

### 5.1 Conclusions

The validity of the proposed procedure was demonstrated through a case study of 1.5 km long road stretch and one signalized intersections with SUMO. Major conclusions, which merge from the thesis, are as following:-

- There is major flow of traffic in Srinagar Highway than the other two side of junctions i.e. Service road facing G11 and Nust. Priority should be given to main highway but at the peak hours 8 am to 10 am in morning and 4 pm to 6 pm at evening there is increase in traffic volume to the service road of Nust.
- Complete follow up of Traffic condition concludes that there is continuous increase in traffic volume which is leading to traffic congestion. Data for last 5 years showed an average addition of 62,000 vehicles each year to city's overall vehicle fleet.
- Passenger cars had the largest vehicle share followed by motorcycles, taxis and vans.

- Simulation based results showed that actuated traffic light control perform better than fixed traffic light control. Actuated traffic light control was found to reduce intersection delays with up to 27% relative to fixed traffic light control. Actuated traffic light control cater fluctuations in traffic flow better than fixed traffic light control.
- The microscopic calibration and validation results show, that simulation platform such as SUMO based on the psycho-physical car-following model can reproduce traffic flow very realistically under different real-world conditions. But for this it is necessary to adapt the model to the local traffic situation i.e. National traffic regulations, driving styles and characteristics of the traffic mix must be taken into account.

### 5.2 **Recommendations**

**Data Collection:** In this research Probabilistic distributions are used to generate traffic data by using population parameters i.e: Inhabitants, households, children age limit, car rate and so on as input. On the basis of that data simulations are performed. The calibration of microscopic simulation models depends on the quality of the observed data. A major concern in modelling and simulation is error inherent in the generation of input data which can be improved by using real time data.

**Purchase of Data Collection Equipment:** Purchase of modern data collection equipment like, electronic counters, GPS, video cameras and speed guns etc. can improve the data collection process and authenticity of data.

**Training on Simulation Packages:** Considerable amount of time was consumed to learn the simulation software SUMO with considerable difficulty. It is recommended that two to three simulation packages be purchased and training on these be arrange t get full access full features of SUMO which will produce better results.

**Recommended area of Application for Pakistan:** Due to their cost effectiveness, risk free nature and high speed benefits, the use of Microscopic traffic simulation models is highly recommended to be used in Pakistan. The recommended area of application is as following:

- Transportation system design
- Traffic operations

• Management alternatives evaluation
## Bibliography

- [1] [n.d.]. Activitygen. https://sumo.dlr.de/docs/Demand/ Activity-basedDemandGeneration.html. Accessed: 2021-06-08.
- [2] [n.d.]. Netedit. https://sumo.dlr.de/docs/netconvert.html. Accessed: 2021-06-08.
- [3] [n.d.]. OpenStreetMap. https://www.openstreetmap.org/export#map= 15/33.6571/73.0008. Accessed: 2021/6/14.
- [4] [n.d.]. The History of Traffic Lights. https://stoneacre.co.uk/blog/ the-history-of-traffic-lights. Accessed: 2021-06-08.
- [5] [n.d.]. Trafficlight. https://sumo.dlr.de/docs/Simulation/Traffic Lights.html. Accessed: 2021-06-08.
- [6] Muhammad Adeel. 2010. Methodology for identifying urban growth potential using land use and population data: A case study of Islamabad Zone IV. *Procedia Environmental Sciences* 2 (2010), 32–41.
- [7] Aminah Hardwan Ahmed. 2018. A review of adaptive intelligent traffic control systems.
  *Journal for Research on Business and Social Science (ISSN (Online) 2209-7880)* 1, 1 (2018).
- [8] Harpreet Kaur Amanjot Kaur, Dr. Mohita Garag. 2019. Review of traffic management control techniques. 7 (2019).
- [9] Khosravi A. Creighton D. Araghi, S. 2015. A review on computational intelligence methods for controlling traffic signal timing. *Expert Systems with Applications* (2015), 1538– 1550.
- [10] Rosa Arce-Ruiz, Neus Baucells, and Concepcion Moreno Alonso. 2016. Smart Mobility in Smart Cities. https://doi.org/10.4995/CIT2016.2016.3485.

- [11] Yuta Asano, Nobuyasu Ito, Hajime Inaoka, Tetsuo Imai, and Takeshi Uchitane. 2015. Traffic simulation of Kobe-city. In *Proceedings of the International Conference on Social Modeling and Simulation, plus Econophysics Colloquium 2014.* Springer, Cham, 255–264.
- [12] Ali Bazghandi. 2012. Techniques, advantages and problems of agent based modeling for traffic simulation. *International Journal of Computer Science Issues (IJCSI)* 9, 1 (2012), 115.
- [13] Moshe Ben-Akiva, Michel Bierlaire, Haris Koutsopoulos, and Rabi Mishalani. 1998. Dyna-MIT: a simulation-based system for traffic prediction. In DACCORD short term forecasting workshop. Citeseer, 1–12.
- [14] Alberto Bull, NU CEPAL, et al. 2003. Traffic Congestion: The Problem and how to Deal with it. ECLAC.
- [15] Benjamin Ng Chan-Tong Lam, Hanyang Gao. 2017. A Real-Time Traffic Congestion Detection System Using On-Line Images. 17th IEEE International Conference on Communication Technology (2017), 195–216.
- [16] Bo Chen and Harry H Cheng. 2010. A review of the applications of agent technology in traffic and transportation systems. *IEEE Transactions on intelligent transportation systems* 11, 2 (2010), 485–497.
- [17] Lara Codeca, Raphaël Frank, and Thomas Engel. 2015. Luxembourg sumo traffic (lust) scenario: 24 hours of mobility for vehicular networking research. In 2015 IEEE Vehicular Networking Conference (VNC). IEEE, 1–8.
- [18] Kevin Pope Daniel Hoornweg. 2017. Population predictions for the world's largest cities in the 21st century. *Environment urbanization* (2017), 195–216.
- [19] JN Darroch, Gordon Frank Newell, and RWJ Morris. 1964. Queues for a vehicle-actuated traffic light. Operations Research 12, 6 (1964), 882–895.
- [20] Juan de Dios Ortúzar and Luis G Willumsen. 2011. Modelling transport. John wiley & sons.
- [21] Samah El-Tantawy, Baher Abdulhai, and Hossam Abdelgawad. 2013. Multiagent reinforcement learning for integrated network of adaptive traffic signal controllers (MARLIN-ATSC): methodology and large-scale application on downtown Toronto. *IEEE Transactions on Intelligent Transportation Systems* 14, 3 (2013), 1140–1150.

- [22] Jakob Erdmann. 2015. SUMO's lane-changing model. In Modeling Mobility with Open Data. Springer, 105–123.
- [23] Yasir Faiz, Md Tufail, M Tayyeb Javed, MM Chaudhry, et al. 2009. Road dust pollution of Cd, Cu, Ni, Pb and Zn along islamabad expressway, Pakistan. *Microchemical Journal* 92, 2 (2009), 186–192.
- [24] Martin Fellendorf. 1994. VISSIM: A microscopic simulation tool to evaluate actuated signal control including bus priority. In 64th Institute of Transportation Engineers Annual Meeting, Vol. 32. Springer, 1–9.
- [25] B.J.; Cunningham C.M. Findley, D.J.; Schroeder and T.H. Brown. 2016. Highway Engineering; Planning, Design, and Operations. (2016).
- [26] Maheen Firdous, Fasih Ud Din Iqbal, Nouman Ghafoor, Nauman Khalid Qureshi, and Noman Naseer. 2019. Traffic light control system for four-way intersection and t-crossing using fuzzy logic. In 2019 IEEE International Conference on Artificial Intelligence and Computer Applications (ICAICA). IEEE, 178–182.
- [27] Bilal Ghazal, Khaled Khatib, Khaled Chahine, and Mohamad Kherfan. 2016. Smart traffic light control system. 140–145. https://doi.org/10.1109/EECEA.2016. 7470780
- [28] Robert L Gordon, Robert A Reiss, Herman Haenel, E Case, Robert L French, Abbas Mohaddes, Ronald Wolcott, et al. 1996. *Traffic control systems handbook*. Technical Report. United States. Federal Highway Administration. Office of Technology Applications.
- [29] Peter Hidas. 2002. Modelling lane changing and merging in microscopic traffic simulation. *Transportation Research Part C: Emerging Technologies* 10, 5-6 (2002), 351–371.
- [30] J.; Oxley J. Johnson, M.; Charlton and S. Newstead. 2013. Why do cyclist infringe at red lights? An investigation of Australian Cyclists' reasons for red light infringement. *Accident Analysis and Prevention* (2013), 840–847.
- [31] Anurag Kanungo, Ayush Sharma, and Chetan Singla. 2014. Smart traffic lights switching and traffic density calculation using video processing. In 2014 Recent Advances in Engineering and Computational Sciences (RAECS). 1–6. https://doi.org/10.1109/ RAECS.2014.6799542

- [32] M. Karim and A. Nasar. [n.d.]. Migration patterns and differentials in Pakistan: Based on the analysis of 1998 census data. In *Population of Pakistan: An Analysis of 1998 Population and Housing Census*, A. Kemal, M. Irfan, and N. Mehmood (Eds.). Pakistan Institute of Development Economics (PIDE), UNFPA, Islamabad.
- [33] A. Kemal, M. Irfan, and N. Mahmood. [n.d.]. Population of Pakistan: An analysis of 1998 Population and Housing Census. Pakistan Institute of Development Economics (PIDE) UNFPA, Islamabad(ICT.
- [34] Abhirup Khanna. 2018. Intelligent Traffic Management System for Smart Cities.
- [35] Peter Koonce and Lee Rodegerdts. 2008. Traffic signal timing manual. Technical Report. United States. Federal Highway Administration.
- [36] I L. Z. Marek L. I P. T. Koukol, M. 2015. Fuzzy logic in traffic engineering : A review on signal control. *Mathematical Problems in Engineering* 2015 (2015), 1–14.
- [37] Daniel Krajzewicz, Elmar Brockfeld, Jürgen Mikat, Julia Ringel, Christian Rössel, Wolfram Tuchscheerer, Peter Wagner, and Richard Wösler. 2005. Simulation of modern traffic lights control systems using the open source traffic simulation SUMO. In *Proceedings of the 3rd Industrial Simulation Conference 2005.* EUROSIS-ETI, 299–302.
- [38] Daniel Krajzewicz, Jakob Erdmann, Michael Behrisch, and Laura Bieker. 2012. Recent development and applications of SUMO-Simulation of Urban MObility. *International journal* on advances in systems and measurements 5, 3&4 (2012).
- [39] Daniel Krajzewicz, Georg Hertkorn, Julia Ringel, and Peter Wagner. 2005. Preparation of digital maps for traffic simulation; part 1: Approach and algorithms. In *Proceedings of the 3rd Industrial Simulation Conference 2005.* EUROSIS-ETI, 285–290.
- [40] Daniel Krajzewicz, Georg Hertkorn, Christian Rössel, and Peter Wagner. 2002. SUMO (Simulation of Urban MObility)-an open-source traffic simulation. In Proceedings of the 4th middle East Symposium on Simulation and Modelling (MESM20002). 183–187.
- [41] Stefan Krauß. 1998. Microscopic modeling of traffic flow: Investigation of collision free vehicle dynamics. (1998).
- [42] Freddy Kurniawan, Denny Dermawan, Okto Dinaryanto, and Mardiana Irawati. 2014. Pre-Timed and Coordinated Traffic Controller Systems Based on AVR Microcontroller.

TELKOMNIKA (Telecommunication Computing Electronics and Control) 12 (12 2014), 787. https://doi.org/10.12928/telkomnika.v12i4.497

- [43] Xiaoyuan Liang, Xunsheng Du, Guiling Wang, and Zhu Han. 2019. A Deep Reinforcement Learning Network for Traffic Light Cycle Control. *IEEE Transactions on Vehicular Technology* 68, 2 (2019), 1243–1253. https://doi.org/10.1109/TVT.2018. 2890726
- [44] Pablo Alvarez Lopez, Michael Behrisch, Laura Bieker-Walz, Jakob Erdmann, Yun-Pang Flötteröd, Robert Hilbrich, Leonhard Lücken, Johannes Rummel, Peter Wagner, and Evamarie Wießner. 2018. Microscopic Traffic Simulation using SUMO, In The 21st IEEE International Conference on Intelligent Transportation Systems. IEEE Intelligent Transportation Systems Conference (ITSC). https://elib.dlr.de/124092/
- [45] JYK Luk. 1984. Two traffic-responsive area traffic control methods: SCAT and SCOOT. *Traffic engineering & control* 25, 1 (1984).
- [46] Anu Maria. 1997. Introduction to modeling and simulation. In Proceedings of the 29th conference on Winter simulation. 7–13.
- [47] Charalampos Marmaras, Erotokritos Xydas, and Liana Cipcigan. 2017. Simulation of electric vehicle driver behaviour in road transport and electric power networks. *Transportation Research Part C: Emerging Technologies* 80 (2017), 239–256. https://doi.org/ 10.1016/j.trc.2017.05.004
- [48] Muhammad Masood, Muhammad Khan, and Hasnain Naqvi. 2011. Transportation Problems in Developing Countries Pakistan: A Case-in-Point. International Journal of Business and Management 6 (10 2011). https://doi.org/10.5539/ijbm. v6n11p256
- [49] William R McShane and Roger P Roess. 1990. Traffic engineering.
- [50] A. J. Miller. 1963. Settings for fixed-cycle traffic signals. The Journal of the Operational Research Society (1963), 373–386.
- [51] MUTCD. 2007. Manual on Uniform Traffic Devices for Streets and Highways. United States Department of Transportation (2007).
- [52] Gunasekaran Muthumanickam and Gopalakrishnan Balasubramanian. 2017. A traffic congestion control in urban areas with vehicle-infrastructure communications. In

2017 International Conference on Energy, Communication, Data Analytics and Soft Computing (ICECDS). 2970–2974. https://doi.org/10.1109/ICECDS.2017. 8390000

- [53] R. Bretherton N. Hounsell, J. Landles and K. Gardner. [n.d.]. Intelligent systems for priority at traffic signals in London: the INCOME project. *International Conference on Road Transport Information and Control* ([n.d.]), 90–94.
- [54] Katsuhisa Ohno and Hisashi Mine. 1973. Optimal traffic signal settings—II. A refinement of Webster's method. *Transportation Research* 7, 3 (1973), 269–292. https://doi. org/10.1016/0041-1647(73)90018-X
- [55] Byungkyu Park, Jongsun Won, Michael A Perfater, et al. 2006. Microscopic simulation model calibration and validation handbook. Technical Report. Virginia Transportation Research Council.
- [56] Daniel Pavleski, Daniela Koltovska Nechoska, and E. Ivanjko. 2018. Evaluation of Adaptive And Fixed Time Traffic Signal Strategies: Case Study of Skopje.
- [57] Andreas Pell, Andreas Meingast, and Oliver Schauer. 2017. Trends in Real-time Traffic Simulation. *Transportation Research Procedia* 25 (2017), 1477–1484. https:// doi.org/10.1016/j.trpro.2017.05.175 World Conference on Transport Research - WCTR 2016 Shanghai. 10-15 July 2016.
- [58] A. Preethi, P.; Varghese and R. Ashalatha. 2016. Modelling Delay at Signalised Intersections under Heterogeneous Traffic Conditions. 11th Transport Planning and Implementation Methodologies for Developing Countries (2016).
- [59] Muhammad Qadir, Jamshaid Zaidi, Sheikh Ahmad, Asad Gulzar, Muhammad Yaseen, Sadia Atta, and Asma Tufail. 2012. Evaluation of trace elemental composition of aerosols in the atmosphere of Rawalpindi and Islamabad using radio analytical methods. Applied radiation and isotopes : including data, instrumentation and methods for use in agriculture, industry and medicine 70 (03 2012), 906–10. https://doi.org/10.1016/j. apradiso.2012.02.047
- [60] M. Besley R. Akçelik and E. Chung. 2001. An evaluation of scats master isolated control. (2001).

- [61] B.S. Reddy and N.V.H. Reddy. 2016. Signal Design for T-intersection by using Webster's Method in Nandyal Town, Kurnool District of Andhra Pradesh. International Research Journal of Engineering and Technology (2016), 1124–1131.
- [62] Simões M. de L. de O. Ribeiro, I. M. 2015. The fully actuated traffic control problem solved by global optimization and complementarity. Engineering Optimization. (2015), 199–212.
- [63] S.hariharagopalan R.keerthi. 2016. A survey on various traffic management schemes for traffic clearance, stolen vehicle and emergency vehicle. *International Journal of Emerging Technology in Computer Science Electronics (IJETCSE)* (2016).
- [64] NAGUI ROUPHAIL15, ANDRZEJ TARKO16, and JING LI17. 1992. Traffic flow at signalized intersections. (1992).
- [65] O. Masoud R. Janardan S. Atev, H. Arumugam and N. P. Papanikolopoulos. 2005. A visionbased approach to collision prediction at traffic intersections. *IEEE Transactions on Intelligent Transportation Systems* 6 (2005), 416–423.
- [66] Munir H Shah and N Shaheen. 2007. Statistical analysis of atmospheric trace metals and particulate fractions in Islamabad, Pakistan. *Journal of hazardous materials* 147, 3 (2007), 759–767.
- [67] Pakistan Bureau Statistics. [n.d.]. Provisional Summary Results of 6th Population and Housing Census. Pakistan Bureau of Statistics, Ministry of Statistics, Islamabad.
- [68] Johannes Stübinger and Lucas Schneider. 2020. Understanding Smart City—A Data-Driven Literature Review. Sustainability 12, 20 (2020), 8460.
- [69] Wasim Hashmi Syed, Ansar Yasar, Davy Janssens, and Geert Wets. 2014. Analyzing the real time factors: which causing the traffic congestions and proposing the solution for Pakistani City. *Procedia Computer Science* 32 (2014), 413–420.
- [70] Martin Treiber and Arne Kesting. 2013. Traffic flow dynamics. Traffic Flow Dynamics: Data, Models and Simulation, Springer-Verlag Berlin Heidelberg (2013).
- [71] Intikhab Ulfat, Farrukh Javed, FA Abbasi, F Kanwal, A Usman, M Jahangir, and F Ahmed.
  2012. Estimation of solar energy potential for Islamabad, Pakistan. *Energy Procedia* 18 (2012), 1496–1500.
- [72] F.V. Webster and B.M. Cobbe. 1966. Traffic Signals. Road Research Laboratory (1966).

- [73] F. V. Webster. 1958. TRAFFIC SIGNAL SETTINGS. Road Research Laboratory Technical Paper 39 (1958), 1–44.
- [74] Xiao-Feng Xie, Yiheng Feng, Stephen F Smith, and K Larry Head. 2014. Unified route choice framework: Specification and application to urban traffic control. *Transportation Research Record* 2466, 1 (2014), 105–113.
- [75] Xiao-Feng Xie, Stephen F Smith, Liang Lu, and Gregory J Barlow. 2012. Schedule-driven intersection control. *Transportation Research Part C: Emerging Technologies* 24 (2012), 168–189.
- [76] C.Y.D Yang and W.G. Najm. 2007. Examining Driver Behaviour using Data gathered from Red light Photo Enforcement Cameras. *Journal of Safety Research* (2007), 311–321.
- [77] Irfan Younas, Muhammad Ilyas, and Riaz Ali. 2008. A traffic advisory system for Islamabad. *Communications of the IBIMA* 3 (2008), 56–61.
- [78] Ilsoo Yun and Byungkyu Park. 2003. Stochastic optimization method for coordinated actuated signal systems. Center for Transportation Studies, University of Virginia.
- [79] A. Zakariya and S. Rabia. 2016. Estimating the minimum delay optimal cycle length based on a time-dependent delay formula. *Alexandria Engineering Journal* (2016).
- [80] Yingying ZHANG, Xumei CHEN, Xiao ZHANG, Guohua SONG, Yanzhao HAO, and Lei YU. 2009. Assessing Effect of Traffic Signal Control Strategies on Vehicle Emissions. Journal of Transportation Systems Engineering and Information Technology 9, 1 (2009), 150–155. https://doi.org/10.1016/S1570-6672(08)60050-1
- [81] Xing Zheng and Will Recker. 2013. An adaptive control algorithm for traffic-actuated signals. Transportation Research Part C: Emerging Technologies 30 (2013), 93–115. https: //doi.org/10.1016/j.trc.2013.02.007